



File No.: 04-1000-20-2023-543

September 25, 2023

s.22(1)

Dear s.22(1)

Re: Request for Access to Records under the Freedom of Information and Protection of Privacy Act (the "Act")

I am responding to your request of August 31, 2023 under the *Freedom of Information and Protection of Privacy Act* for:

Record of joint operating agreement between the Vancouver Park Board and the Killarney Community Centre Society as of August 30, 2023.

All responsive records are attached.

Under section 52 of the Act, and within 30 business days of receipt of this letter, you may ask the Information & Privacy Commissioner to review any matter related to the City's response to your FOI request by writing to: Office of the Information & Privacy Commissioner, info@oipc.bc.ca or by phoning 250-387-5629.

If you request a review, please provide the Commissioner's office with: 1) the request number (#04-1000-20-2023-543); 2) a copy of this letter; 3) a copy of your original request; and 4) detailed reasons why you are seeking the review.

Yours truly,

[Signed by Cobi Falconer]

Cobi Falconer, MAS, MLIS, CIPP/C Director, Access to Information & Privacy <u>cobi.falconer@vancouver.ca</u> 453 W. 12th Avenue Vancouver BC V5Y 1V4 If you have any questions, please email us at foi@vancouver.ca and we will respond to you as soon as possible. Alternatively, you can call the FOI Case Manager at 604-871-6584.

Encl. (Response Package)

:dl



LAW DEPARTMENT

File No.: 18-01895-007

June 18, 2018

MEMORANDUM

TO:

Janice MacKenzie, City Clerk

CC;

Donnie Rosa, Park Board (without enclosure)

FROM:

Karin Grubb, Solicitor, Law Department

SUBJECT:

Joint Operating Agreement dated May 14, 2018 between The Vancouver Board of Parks and Recreation and Killarney Community Centre Society and Mutual Release dated May 15, 2018 among Killarney Community Centre Society and The Vancouver Board of Parks and Recreation of The

City of Vancouver and the City Of Vancouver

Enclosed please find the above-noted Agreement for filing. Please note the following:

TYPE OF AGREEMENTS	One (1) signed Joint Operating Agreement and One (1) signed Mutual Release
DATE OF AGREEMENTS (if this date is execution date and there is more than one date, use the latest date as the date of the agreement)	May 14, 2018 - <i>Joint Operating Agreement</i> May 15, 2018 - <i>Mutual Release</i>
PARTIES (complete names)	The Vancouver Board of Parks and Recreation and Killarney Community Centre Society – <i>Joint Operating Agreement</i>
	Killarney Community Centre Society and The Vancouver Board of Parks and Recreation of The City of Vancouver and the City Of Vancouver – <i>Mutual Release</i>
CIVIC ADDRESS (no abbreviations - must be searchable)	6260 Killarney Street
LEGAL DESCRIPTION (no abbreviations - must be searchable)	Lot 17 except part in Explanatory Plan 19208 Block A District Lot 337 Plan 11401

{01017865v1}

Page 1

City of Vancouver, Law Department 453 West 12th Avenue Vancouver, BC V5Y 1V4 Canada Telephone: (604) 873-7512

Telephone: (604) 873-7512 Fax: (604) 873-7445 Site/Delivery Address: 401-515 West 10th Avenue Vancouver, BC V5Z 4A8 Canada



EXPIRY DATE (indicate "N/A" if there is no expiry date)	Renewal Date: December 31, 2028 Expiry Date (if renewed): December 31, 2034
REMINDER DATE (if there are no expiry/retention dates, THIS MUST BE FILLED IN - the lawyer can help to determine when this agreement could possibly be sent to Records - City Clerks will use this date to remind us to review the agreement to determine whether it can be sent to Records)	
RETENTION DATE (if there is an expiry date, fill in the date that the City Clerk may send this document to Records - usually one year after expiry date)	December 31, 2028
STAFF INVOLVED: (this may be more than one person) make sure you indicate the person's name, department (including division) and telephone number	Karin Grubb Solicitor, Law Department Phone: (604) 871-6913 Donnie Rosa Park Board Phone: (604) 257-8638

FILE NUMBER	
(Law Department file number)	18-01895-007

Karin Grubb

KAG:rp Attachment

PARK BOARD - COMMUNITY CENTRE ASSOCIATION JOINT OPERATING AGREEMENT

BETWEEN

THE VANCOUVER BOARD OF PARKS AND RECREATION

AND

KILLARNEY COMMUNITY CENTRE SOCIETY

TABLE	OF CONTENTS	1	3.4	ART INSTALLATIONS AND MURALS	24
		1	3.5	VEHICLES	24
1. D	EFINITIONS4	14.	FIN	ANCE	24
2. LI	EGAL RELATIONSHIP9	1	4.1	REVENUE	24
		1	4.2	Expenses	25
3. T	ERM OF AGREEMENT9	1	4.3	BUDGETS AND RECORDS	27
3.1	TERM9	1	4.4	GRANTS	29
3.2	Renewal9	1	4.5	OPERATIONS FEE	30
3.3	PROCESS AT END OF AGREEMENT10	1	4.6	INSURANCE	31
4. A	SSOCIATION GOVERNANCE10	1	4.7	MISCELLANEOUS OPERATIONS	
4.1	GOVERNANCE COVENANTS10	15.	COI	MMUNICATIONS AND PLANNING	31
4.2	GOVERNANCE DEFAULTS	1	5.1	COMMUNICATIONS BETWEEN PARTIES	31
		1.	5.2	SYSTEM-WIDE PLANNING AND COMMUNIC	
	ENERAL OBLIGATIONS AND ROLES OF THE	Sı	ESSION	ıs 32	
PARTIE	S11				
5.1	MUTUAL OBLIGATIONS11	16.		ORMATION AND BUSINESS SYSTEMS,	
5.2	ROLE OF THE PARK BOARD	TECH	INOL	OGY & TELECOMMUNICATIONS	32
5.3	ROLE OF THE ASSOCIATION	1	6.1	REGISTRATION SYSTEM	32
		1	6.2	OWNERSHIP OF SYSTEMS	
6. P	OLICIES AND PROCEDURE13	1	6.3	SECURITY SYSTEMS	
6.1	CORPORATE AND GOVERNANCE POLICIES	1	6.4	MANAGEMENT OF DATA	
6.2	CHANGES TO PUBLIC POLICY14	1	6.5	USE OF BUSINESS SYSTEMS AND INFORMAT	ION
		Ti	ECHNO	DLOGY	
7. P	ROGRAMMING14				
7.1	Access to Programming and the Jointly	17.		NERSHIP OF COMMUNITY CENTRE AS	SETS
•	ATED FACILITIES		34		
7.2	PROGRAMMING RESPONSIBILITIES	1	7.1	COMMUNITY CENTRE FACILITY	34
7.3	SYSTEM-WIDE PROGRAMS	1	7.2	EQUIPMENT AND ASSETS	35
7.4	QUALITY CONTROL AND EVALUATION	40	D.0		
		18.	DIS	PUTE RESOLUTION	35
8. A	SSOCIATION MEMBERSHIP17	18	8.1	DISPUTE RESOLUTION PROCESS	35
9. U	SE ALLOCATION FOR JOINTLY OPERATED	18	8.2	ARBITRATION	36
FACILIT	IES 17	19.	BRA	AND AND BRAND PROTECTION	37
10. C	HILDCARE18	20.		ANGES TO JOINTLY OPERATED FACILIT	
11. A	FFILIATED GROUPS18				
			0.1	Additions to Entire Facility	
12. H	UMAN RESOURCES19	20	0.2	REDUCTION IN JOINTLY OPERATED FACILITIES	Es 37
12.1	Association Personnel and Volunteers 19	20	0.3	REPLACEMENT OF JOINTLY OPERATED FACIL	ITIES
12.1				37	
12.2	20	20	0.4	CLOSURE OF JOINTLY OPERATED FACILITIES	38
12.3	Park Board Personnel20	21.	TER	MINATION OR NON-RENEWAL	38
12.4		-		B	,_
12.5	HIRING DECISIONS21		1.1	BREACH BY ASSOCIATION AND PARK BOARD	
12.6				O TERMINATE	
			1,2	BREACH BY PARK BOARD AND ASSOCIATION	-
13. R	ENOVATIONS AND MAINTENANCE22			O TERMINATE	
13.1	RENOVATIONS, UPGRADES AND MAINTENANCE OF	22.	NO.	rices	39
	LY OPERATED FACILITIES22	23.	TER	MINATION OF PRIOR JOA	40
13.2	Maintenance of Other Assets23				
13.3	PLANNING FOR SURROUNDING AREAS	24.	MIS	CELLANEOUS	40

{00704256v1}

24.1	GOVERNING LAWS	40
24.2	ENUREMENT	40
24.3	SEVERABILITY	40
24.4	Non-Derogation	41
24.5	TIME OF THE ESSENCE	41
24.6	CAPTIONS	41
24.7	INTERPRETATION	41
24.8	WAIVER	
24.9	ENTIRE AGREEMENT	
24.10	APPENDIX C PREVAILS	42
24.11	ASSIGNMENT	
24.12	INDEPENDENT LEGAL ADVICE	
24.13	COLLECTIVE AGREEMENTS	
24.14	CITY AS PUBLIC BODY	
24.15	PERMANENT PUBLIC PARK	43
APPENDIX	A	44
APPENDIX	В	45
APPENDIX	C ERROR! BOOKMARK NOT DEFI	NED.
APPENDIX	D	55
APPENDIX	E	57

PARK BOARD - COMMUNITY CENTRE ASSOCIATION AGREEMENT

THIS AGREEMENT is made the 14 day of May, 2018

BETWEEN:

The Vancouver Board of Parks and Recreation 2099 Beach Avenue Vancouver, BC V6G 1Z4

(the "Park Board")

AND:

Killarney Community Centre Society, Society Number S0005158 6260 Killarney Street Vancouver, BC V5S 2X7

(the "Association")

WHEREAS:

- A. Each of Vancouver's community centres serves as the hub of its respective neighbourhood linking residents together with resources to encourage positive health outcomes through healthy lifestyles, social connectedness and the ability to contribute to community;
- B. Community centres provide access to quality programming and services to encourage healthy lifestyles through participation in physical, recreational, cultural, educational, social and capacity-building programs for residents of the community, which are responsive to the community's unique needs;
- C. The Park Board is an elected board of commissioners with statutory jurisdiction over Vancouver parks, including the provision of recreational programs and facilities, setting policies and allocating budget resources and the authority to organize, conduct and contract with others for recreational programs;
- D. The City of Vancouver is the sole owner of the assets which are in the possession and control of the Park Board, including designated parks and other land, the community centre buildings and other facilities and improvements located on such lands;
- E. The Association is a valuable community-based society incorporated under the Societies Act with an elected Board of Directors that plays an important role in contributing to the success of the Community Centre Network, including by delivering Programming and services which ensure a healthy, engaged and inclusive community;
- F. While the parties recognize the unique attributes of each community centre and neighbourhood, the Park Board and the Association agree that all residents of the City of Vancouver should have similar opportunities, regardless of gender, age, sexual orientation, ability, race, culture, faith or socio-economic status;

[00704256v1] 3

- G. The parties agree that it is in their mutual interest to work cooperatively to ensure the successful and financially sustainable offering of Programming and services from the Jointly Operated Facilities and to improve community engagement and participation through the Jointly Operated Facilities;
- H. This Agreement replaces the previous joint operating agreement between the parties dated June 1, 1979 (the "Prior JOA"). This updated operating agreement reflects the current working relationship of the parties and is entered into because both parties recognize and appreciate the skills and attributes contributed by the other to the provision of Programming and services to the community. This working relationship reflects a mutual acknowledgement and respect of the parties' distinct roles and shared objectives;
- I. The scope of this Agreement is limited to the Jointly Operated Facilities, and any additional scope (for example, parks and fieldhouses) based on current arrangements will be addressed in the appendices; and
- J. This Agreement sets out the legal relationship between the Park Board and the Association for the operation of the Jointly Operated Facilities. It also clarifies the roles of the parties and enables the Association to provide Programming and services in the Jointly Operated Facilities in support of the overall goals of the Park Board and Association and to the benefit of the residents of the City of Vancouver.

NOW THEREFORE, in consideration of the payment of ten dollars by each party to the other, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged and agreed to, the parties hereto agree as follows:

1. DEFINITIONS

- (a) "ActiveNet" has the meaning ascribed in Section 16.1;
- (b) "Applicable Laws" means all laws, regulations and governmental policies of any governmental authority, including Park Board or City by-laws, as they may be amended or replaced from time to time, and which are applicable to each of the parties' operations in the Jointly Operated Facilities and the roles and responsibilities of each party under this Agreement;
- (c) "Association Board" means the Board of Directors of the Association;
- (d) "Association Equipment" has the meaning ascribed in Section 17.2 of this Agreement:
- (e) "Association Personnel" has the meaning ascribed in Section 12.1 of this Agreement;
- (f) "Authorized Deductions" means deductions for amounts owing by the Association in respect of:
 - (i) 3rd party provider credit, debit card or terminal transaction fees, which are 2.1% as of the Effective Date, and will be reviewed annually and adjusted, if necessary;

- recovering revenue paid to the Association in error due to processing or calculating errors which resulted in the Association receiving revenue which it is not entitled to pursuant to this Agreement (with prior written notification);
- (iii) any applicable Registration System subscription fee on transactions that are processed through the Registration System. The subscription fee is currently fixed at 1% for the term of the agreement (2014-2019) with Active Network Ltd.; and
- (iv) any other deductions to be made from revenue and which are agreed to in writing by the Association;
- (g) "Breaches" means breaches, claims, demands, actions, and costs and expenses incurred in respect thereof;
- (h) "CCAs" means, collectively, the community centre associations which work with the Park Board pursuant to joint operating agreements and provide services in respect of the Community Centre Network;
- (i) "City" means the City of Vancouver, as a municipal corporation;
- (j) "City Networks" means all City owned and leased, wired and wireless, information networks used for City information technology operations, including providing information technology services for the Park Board;
- (k) "Centrally Processed Revenue" means all Association revenue processed through the Registration System, which includes all Facility-Generated Revenue, except for the following excluded items which are not processed through the Registration System: revenue from vending and concessions, and Licensed Childcare optionally excluded as set out in Appendix D(p);
- (l) "Common Spaces" means the interior areas of the Entire Facility (such as hallways and washrooms) used in common by the parties to access and use the various rooms and spaces comprising the Jointly Operated Facilities, in accordance with the current practices of the parties and as further described in Appendix A;
- (m) "Community Centre Network" means the network of community centre facilities in the City of Vancouver operated by the Park Board in conjunction with CCAs pursuant to joint operating agreements;
- (n) "Director of Recreation" means the Director of Recreation of the Park Board, and their successors or those authorized to act in their place;
- (o) "Effective Date" means January 1, 2018, which is the date that this Agreement comes into effect;
- (p) "Entire Facility" means the Lands, the building and facilities commonly known as the Killarney Community Centre, including the Jointly Operated Facilities, and the surrounding or adjacent parks, arenas, pools, playing fields, Park Board fitness facilities and other outdoor recreation improvements or facilities such as,

- without limitation, tennis courts, basketball courts, skateboard parks or field houses within or in the vicinity of the Killarney Community Centre and which are administered by the Park Board;
- "Facility-Generated Revenue" means all revenue generated by the Association with the assistance of the Park Board or through the Association's use of the Jointly Operated Facilities, including from Programming (including at Satellite Locations), room rentals, Association-operated Licensed Childcare, payments due under childcare leases, concessions and vending machines, any Association operated fitness centre and special events held in the Jointly Operated Facilities and any other revenue identified as "facility-generated revenue" in Appendix C; however, "Facility-generated Revenue" expressly excludes grant, bequest, donation and fundraising revenue, interest and investment income, and refunded rental deposits;
- (r) "Fiscal Year" means the 12 month period which is the fiscal year of the Park Board or the Association, respectively, as the context requires. The fiscal year of the Park Board commences on the first day of January in one calendar year and ends on the last day of December in the same calendar year;
- (s) "Fixtures" means items that are part of, or secured or affixed to the Entire Facility (such as by being attached by bolts, screws, plaster etc.), including items the removal of which would do damage or necessitate repair to the remaining components of the Lands or building, including, for example, items such as permanent reader boards, lighting, flooring and affixed carpets, basketball hoops, timing clocks and built in cabinets;
- (t) "General Manager" means the General Manager of the Park Board, and their successors or those authorized to act in their place;
- (u) "Identity Management" means the broad administrative area for information technology uses and access that deals with identifying individuals in a system (such as a country, a network, or an enterprise) and controlling the individual's access to resources within that system by associating user rights and restrictions with the established identity;
- (v) "Indemnity Agreement" means the indemnity agreement between the Association, the Park Board and the City dated May 9, 2000, as it may be amended or replaced from time to time, the current form of which is attached to this Agreement as Appendix E;
- (w) "Input" means the contribution of ideas, information or opinions by the Association to the Park Board in response to a question, issue or action proposed by the Park Board. For the purposes of this agreement, "Input" will be provided as follows:
 - the Park Board will provide the Association with an outline of the matter, including additional information that could reasonably be considered relevant to that question, issue or action;

- (ii) the Park Board will provide the Association with a reasonable period of time, depending on the nature of the matter, in which to consider the information and provide a response on the matter to the Park Board;
- (iii) the Park Board will consider any response received from the Association within such period of time before making any final decision on the matter; however, the final decision will rest with the Park Board unless this Agreement expressly states that a particular matter is to be decided otherwise;
- (x) "IT Assets" has the meaning ascribed in Section 16.2;
- (y) "Jointly Operated Facilities" means the portions of the Entire Facility operated jointly by the Park Board and the Association, as more particularly set out in Appendix A, and which, for the purposes of this Agreement, shall not include surrounding or adjacent parks, arenas, pools, playing fields, Park Board fitness facilities and other outdoor recreation improvements or facilities, including the Outdoor Areas, unless expressly identified in Appendix A;
- (z) "Lands" mean those lands that are commonly known as 6260 Killarney Street and legally described as Lot 17 except part in Explanatory Plan 19208 Block A District Lot 337 Plan 11401;
- (aa) "Licensed Childcare" means regulated programs for children under the Community Care and Assisted Living Act, SBC 2002 c. 25 (as may be amended or replaced from time to time), and which may include infant toddler daycare, preschool programs or before or after school programs;
- (bb) "OneCard" means the universal access card provided by the Park Board onto which access products are loaded, and which enables access to products and services at all centres within the Community Centre Network;
- (cc) "Operations Fee" means a fee paid by the Association to the Park Board to be used to increase equity and/or achieve Public Policy goals across the Community Centre Network;
- (dd) "Outdoor Areas" mean the parks surrounding or adjacent to the building(s) known as the Killarney Community Centre, playing fields and other outdoor recreation improvements or facilities, such as tennis courts, basketball courts, skateboard parks, plazas, playgrounds or field houses which are administered by the Park Board and used for Programming;
- (ee) "Park Board" means the Vancouver Board of Parks and Recreation;
- (ff) "Park Board Personnel" means employees or contractors retained by the City to provide services to or on behalf of the Park Board;
- (gg) "Prior JOA" has the meaning ascribed in Recital H;
- (hh) "Programming" means the physical, recreational, cultural, educational, social and capacity-building programs and other services offered to members of the public through registered or unregistered programs administered and run in or

- through the Jointly Operated Facilities, including the System-wide Programs, but which, for greater certainty, does not include Licensed Childcare;
- (ii) "Public Policy" means policy, priorities and objectives adopted or determined by the elected Park Board Commissioners that pertain to access, operations and use of the overall recreation system, including the Community Centre Network, or which may address statutory or regulatory compliance, risk management, system-wide financial issues, sustainability goals for the recreation system, or which are aimed at ensuring better health or quality of life outcomes for the residents of the City of Vancouver;
- (jj) "Recreation Supervisor" means the recreation supervisor at the Jointly Operated Facilities who is employed for the Park Board, and their successors or those authorized to act in their place;
- (kk) "Registration System" means the common recreation and registration management system owned and operated by the City and the Park Board that is used to process, record and store information for various registration functions at or through the Jointly Operated Facilities. The registration system currently in use is ActiveNet, but it may be replaced or substituted with another system or systems from time to time, at the discretion of the Park Board;
- (II) "Renewal Term" has the meaning ascribed in Section 3.2;
- (mm) "Retained Earnings" means the unspent revenue accumulated by the Association over time from Facility-Generated Revenue;
- (nn) "Satellite Locations" means locations that are not part of the Jointly Operated Facilities and are not otherwise owned or leased by the Park Board, and where the Association offers Programming or services that are registered for or offered through the Jointly Operated Facilities or are being provided with the assistance of the Park Board;
- (00) "Societies Act" means the Societies Act, SBC 2015, c.18, as may be amended or replaced from time to time;
- (pp) "Staffing Cost Recovery Payment" means the amount to be paid by the Association to the Park Board for the full cost of employment of certain Park Board Personnel at the Jointly Operated Facilities who are performing work as requested by the Association to provide additional support for Programming or Association services, including the costs commonly referred to as "Group 1";
- (qq) "System-wide Programs" means local, area-based or city-wide Programming offered within and across the Community Centre Network to provide key developmental, health, cultural or social benefits to the city-wide community or a specific sub-population of the city-wide community, and which programs are designated by the Park Board, as further described in Section 7.3;
- (rr) "System-wide Planning Session" has the meaning ascribed in Section 15.2;

- (ss) "Term" means the term of this Agreement set out in 3.1, as may be extended or renewed pursuant to 3.2; and
- (tt) "Vending Contracts" has the meaning ascribed in Section 14.1(b).

2. LEGAL RELATIONSHIP

The Park Board and the Association are independent contracting bodies, not legal partners nor joint employers, nor are they in a landlord-tenant relationship. This Agreement is being entered into between the parties to create certainty around the roles and obligations of the parties with respect to their respective operations at the Jointly Operated Facilities and to permit the Association's use of and operations at the Jointly Operated Facilities, including for the provision of Programming and services to the public through the Jointly Operated Facilities for the benefit of the residents of City of Vancouver. Through this Agreement, once fully executed, the Park Board is granting to the Association the rights set out herein, including the right to non-exclusive use and occupation of the Jointly Operated Facilities and use of the Common Spaces, on and subject to the terms and conditions set out in this Agreement.

3. TERM OF AGREEMENT

3.1 Term

This Agreement is for a term of ten (10) years, commencing on the Effective Date. For greater certainty, provisions for termination or adjustment of this Agreement are included in Articles 20 and 21.

3.2 Renewal

At the end of the initial term, the parties shall have one (1) option to renew this Agreement for a renewal term of five (5) years (the "Renewal Term"). There shall be no further options to renew. No later than six (6) months prior to the expiry of the initial term, either party may provide written notification to the other of that party's election not to renew the Agreement, for any reason, and the Agreement will automatically expire at the end of that 10 year term. Absent such written notification from either party, the Agreement will automatically renew at the end of the initial 10 year term.

If either party wishes to amend any terms or conditions of this Agreement prior to the commencement of the Renewal Term, that party must give the other party written notice of such request, including a description of the desired amendments, at least 6 months prior to the expiration of the initial term. The process for considering desired amendments to the body of this Agreement must be established by the Park Board Commissioners and any changes to the body of this Agreement must be approved by the Park Board Commissioners in a public forum.

If amendments are not finalized prior to the commencement of the Renewal Term, this Agreement will stay in effect un-amended until such time as the amendments are approved by the Park Board Commissioners, or until the end of the Renewal Term, whichever is first.

Desired amendments to Appendix A or Appendix C to this Agreement may be made at any time during the Term upon mutual agreement of the Park Board and the Association.

{00704256v1}

3.3 Process at End of Agreement

At the end of the Term, including any elected Renewal Term, to a maximum of 15 years, this Agreement will terminate, together with any rights of the Association to use or occupy the Jointly Operated Facilities.

If both parties wish to continue their working relationship beyond the end of the Term, then the Park Board and the Association will, in good faith, meet to review their working relationship and to identify any changes to the roles and responsibilities of the parties or to the operations at the Jointly Operated Facilities. If acceptable to both parties, the parties may elect to enter into a new amended agreement reflecting such changes, which may, if appropriate, be substantially based on the form of this Agreement.

In the event of non-renewal of this Agreement or if either party elects it does not wish to continue their working relationship at the end of the Term, all outstanding payments between the parties will become immediately due and payable and subject to withholding, and the parties will work cooperatively to transition operations to the Park Board or its designate to ensure minimal disruption to the public.

4. ASSOCIATION GOVERNANCE

4.1 Governance Covenants

The Association covenants and agrees that:

- (a) it will conduct itself at all times in accordance with its constitution and by-laws and the requirements of the Societies Act;
- (b) as it pertains to activities carried out under this Agreement, the Association will adopt and adhere to appropriate good governance policies;
- to prevent a conflict of interest, the Association shall request any board member who is an elected City or Park Board official, or who has formally indicated a decision to run for election to the City Council or Park Board, to take a leave from or resign from the Board pending the outcome of the election. If the board member refuses to do so, the Association agrees that it shall call a meeting of the Association's voting members to consider whether or not such board member should be removed from the board by special resolution in accordance with the Associations bylaws; and
- (d) it will comply with its policy obligations under Section 6.1.

The Park Board acknowledges and agrees that the Association is a society incorporated under the Societies Act and is bound by the terms of the Societies Act and, notwithstanding any terms of this Agreement, the Park Board will not require the Association to act in any way or commit any act, including the spending of money, that will result in the Association being in breach of the Societies Act.

The Park Board further acknowledges and agrees that the Association may now, or at some time in the future, hold charitable status under the Income Tax Act. Notwithstanding any terms of this Agreement, the Park Board will not require the Association to act in any way or

commit any act, including the spending of money, which will result in the Association being in breach of its obligations as a charitable organization under the Income Tax Act.

If, at any time, it is determined that a section or requirement of this Agreement will result in a breach of the Societies Act or the Income Tax Act, including as a result of a future amendment or revision to the Societies Act or Income Tax Act, the parties will agree on an alternative course of action that is in compliance with the Act and meets the intent of the applicable section or requirement of this Agreement. If the parties cannot agree whether a section or requirement of this Agreement will result in a breach of the Societies Act or on an alternative course of action, then the matter shall be referred to dispute resolution in accordance with Section 18.1, and then to arbitration pursuant to Section 18.2 if still unresolved. If the dispute is referred to arbitration, then the parties shall select an arbitrator who has experience with the Societies Act to determine if the section or requirement of this Agreement in question is a breach of the Societies Act.

4.2 Governance Defaults

To ensure that the use of the Jointly Operated Facilities by the Association for the purposes set out in this Agreement continues to be carried out in the best interests of the public, the parties agree that, if:

- (a) the purpose of the Association is amended so that, at any time, it includes the carrying on of a business for profit or gain;
- (b) the Association makes an application to become a "member-funded society" pursuant to the Societies Act without the prior written approval of the Park Board;
- (c) the purpose of the Association is amended so that it is no longer consistent with benefiting the local and city-wide community and members of the public through the provision of Programming and other services within the Community Centre Network; or
- (d) the Association adopts bylaws and governance procedures that conflict with the terms of this Agreement or do not permit the Association to carry out its roles and responsibilities under this Agreement,

then it shall be considered a default of the Association under this Agreement and the Dispute Resolution Process set out in Section 18.1 of this Agreement shall apply.

5. GENERAL OBLIGATIONS AND ROLES OF THE PARTIES

5.1 Mutual Obligations

In addition to the more specific obligations of the parties set out in this Agreement, the Park Board and the Association agree that:

- (a) in all dealings with each other, they will each communicate respectfully and act in a manner that reflects a mutual acknowledgement and respect of the parties' distinct roles and shared objectives;
- (b) they will each perform their legal and contractual obligations honestly and in good faith; and

(c) they will each comply with all Applicable Laws and will, when appropriate, obtain all necessary permits and licenses as may be required by Applicable Laws.

5.2 Role of the Park Board

The Park Board is an elected board of commissioners with exclusive jurisdiction, possession and control of Vancouver's parks and the authority to organize, conduct and contract with others for recreational programs of all kinds in parks and such other locations as may be approved by the Park Board, pursuant to the provisions of the *Vancouver Charter*, SBC 1953, c.55 and as otherwise authorized by City Council. The Park Board is responsible for:

- (a) development and implementation of Public Policy;
- (b) stewardship of City park lands and facilities;
- (c) stewardship of capital and operating funds allocated by City Council for parks and recreation services; and
- (d) contracting with other parties, including the CCAs, to deliver recreational programs on behalf of the Park Board in accordance with the terms of this Agreement.

In addition to the other responsibilities set out in this Agreement, with respect to operations at the Jointly Operated Facilities the Park Board is more specifically responsible for:

- (e) determining priorities and objectives for implementation of Public Policy at the Jointly Operated Facilities, which includes Public Policy relating to recreation, arts, culture and social programs. The Park Board agrees that any new Public Policy initiatives for implementation at the Jointly Operated Facilities will not expressly contradict the terms of this Agreement, unless agreed to by the parties in advance;
- (f) the staffing of Park Board Personnel at the Jointly Operated Facilities, including scheduling of staff and staff vacations, taking into consideration the budget allocated by the Park Board and operational requirements of the Jointly Operated Facilities;
- (g) directing Park Board Personnel and volunteers;
- (h) developing the Park Board budget (operating and capital) for the Entire Facility and allocating resources as appropriate, taking into consideration any Input provided by the Association in accordance with Section 5.3(f);
- (i) operating, maintaining and constructing improvements or repairs to the Entire Facility in accordance with the Park Board budget;
- (j) undertaking any capital improvements or renovations to, expansion of, or the construction of new facilities at the Entire Facility, and reconstruction and/or relocation of the Entire Facility if the Entire Facility is rebuilt, in the sole discretion of the Park Board; and
- (k) providing and being responsible for all facilities, building systems, supplies, and IT Assets required to operate the Entire Facility, subject to the obligations of the Association set out in this Agreement.

5.3 Role of the Association

The Association is an independent society which is governed by the Societies Act and the Association's constitution and bylaws. In addition to the other responsibilities set out in this Agreement, with respect to its operations at or from the Jointly Operated Facilities, the Association is responsible for:

- (a) developing, delivering, and staffing Programming and other Association services offered in or from the Jointly Operated Facilities;
- (b) the staffing and directing of Association Personnel and volunteers at the Jointly Operated Facilities, including the scheduling of personnel and volunteers and scheduling of vacations taking into consideration the operational requirements of the Jointly Operated Facilities, subject to the provisions of Section 12.2;
- (c) fundraising, recruiting and recognizing volunteers, and advocating for the needs of its community, including with respect to services and facilities;
- (d) engaging with the community and promoting recreation, cultural, social and educational involvement within the community;
- (e) complying with Public Policy determined by the Park Board for implementation at the Jointly Operated Facilities or in respect of Programming or services offered from Jointly Operated Facilities, and cooperating with the Park Board when developing an implementation plan for new Public Policy;
- (f) providing Input to the Park Board on major capital projects and other major initiatives relating to the Jointly Operated Facilities and Outdoor Areas;
- (g) if applicable, the administration, delivery and staffing of any existing Licensed Childcare;
- (h) demonstrating to the community, other CCAs, and the Park Board that its operations are conducted in a financially transparent and responsible manner and that the Association practices good corporate governance, as further described in Section 6.1; and
- (i) ensuring timely payment of all bills and invoices for amounts owed by the Association to the Park Board.

For greater certainty, this Agreement is not intended to restrict or limit the Association from conducting other activities which are outside the scope of this Agreement, including community advocacy.

6. POLICIES AND PROCEDURE

6.1 Corporate and Governance Policies

The Association will adopt and adhere to good governance policies, including policies that address conflict of interest, and succession planning for Association Board members. The Association will conduct regular reviews of its Association Board function and compliance.

The Association will also adopt and adhere to up-to-date corporate policies and standards, appropriate and commensurate with the Association's roles and responsibilities under this Agreement and to the public including, without limitation, policies with respect to occupational health and safety, respectful workplace, risk management and human resources management.

For the purposes of carrying out its roles and responsibilities under this Agreement, the Park Board and Park Board Personnel will adhere to Park Board and City corporate policies and processes, including financial, risk management, information security risk management, privacy and security and conflict of interest, as may be amended from time to time.

The City and the Park Board are subject to certain laws, and have policies in place, that are applicable to the operation of the Entire Facility, such as procurement policies, and the City and the Park Board are party to a collective agreement for labour and employment matters. The Association acknowledges that the Park Board and Park Board Personnel must comply with such laws, policies and agreements, when and where appropriate.

The Park Board will offer optional training sessions for the CCAs on elements of good governance, or other topics as may be agreed to by the Park Board and the CCAs from time to time.

6.2 Changes to Public Policy

The Association acknowledges and agrees that the Park Board is responsible for the development and implementation of Public Policy as it pertains to the Community Centre Network and that Public Policy may be updated or amended from time to time to address, among other things, statutory or regulatory changes, risk management, financial constraints or to ensure better health or quality of life outcomes for the residents of the City of Vancouver.

If Public Policy is changed, amended, revised or updated during the Term of this Agreement or if new Pubic Policy is developed and planned for implementation then, prior to implementing any amendments to Public Policy or new Public Policy that affect the operations of the Association pursuant to this Agreement, the Park Board will clearly explain the context and rationale for the new or amended Public Policy, review any potential impacts to the CCAs and the Community Centre Network and plan for implementation of any changes with Input from the affected CCAs. The Park Board will develop an implementation plan that minimizes disruption or financial impact to the CCAs, to the extent reasonably possible. The Park Board agrees that any new Public Policy initiatives for implementation at the Jointly Operated Facilities will not expressly contradict the terms of this Agreement, unless agreed to by the parties in advance.

7. PROGRAMMING

7.1 Access to Programming and the Jointly Operated Facilities

The Association acknowledges that the scope of Public Policy includes policy regarding access to the Jointly Operated Facilities and Programming or services offered from the Jointly Operated Facilities. Access policy may address issues such as reducing barriers to participation, universal access, leveraging the capacity of the Community Centre Network, and such other matters that the Park Board deems to be of public importance from time to time.

[00704256v1] 14

The current policy with respect to access is attached to this Agreement as Appendix B, and the Association agrees that this policy is acceptable to and binding upon the Association.

7.2 Programming Responsibilities

Subject to Sections 7.1 and 7.3, Programming is the responsibility of the Association, including scheduling, fees, quality control and evaluation. The Association may use Outdoor Areas or other areas of the Entire Facility outside of the Jointly Operated Facilities for the purposes of Programming or special events in accordance with the Park Board permitting process, or with the prior written approval of the Park Board.

The Association's responsibilities for Programming include, without limitation:

- (a) development of the Programming list and schedule;
- (b) production and circulation of promotional materials such as the Programming brochure;
- (c) hiring and contracting with instructors/contractors to deliver Programming, including the System-wide Programs;
- (d) oversight and management of instructors/contractors who are delivering Programming;
- (e) providing and being responsible for adequate supplies to operate Programming;
- (f) oversight, screening and management of any volunteers involved with delivery of Programming and Association services and events;
- (g) assessment of credentials and compliance with any statutory obligations when recruiting instructors, contractors or volunteers;
- (h) ensuring Programming is offered safely with proper risk-management and in a safe, inclusive environment;
- (i) management of Programming-related risks and issues;
- (j) subject to the Park Board's involvement in System-wide Programs, evaluation of Programming in accordance with Section 7.4; and
- the purchase, maintenance, repair and replacement of minor capital equipment and any specialized equipment used or required for Programming or other Association operations (such as fitness centres or Licensed Childcare), as further detailed in Section 14.2(b),

unless any such responsibilities are formally delegated to the Park Board in accordance with Section 12.2.

The Park Board shall provide the Association with available resources, such as demographic information, reports, and best practice information, to support the Association's Programming responsibilities.

7.3 System-wide Programs

System-wide Programs are standardized programs to be offered across the Community Centre Network. System-wide Programs will be designated by the Park Board and implemented in collaboration with the Association. These programs will be based on leading practices and aimed at achieving outcomes that the Park Board has identified as priorities. The Association acknowledges and agrees that System-wide Programs may be offered at the Jointly Operated Facilities in accordance with the following practices:

- (a) the Park Board may designate up to five programs each year as System-wide Programs, which programs may be offered at each community centre in the Community Centre Network:
- (b) the Park Board programming staff will determine if some or all of the System-Wide Programs are appropriate for offering at the Jointly Operated Facilities;
- (c) the Association will be responsible for appropriate scheduling and administration of the System-wide Programs and will include the System-wide Programs in the Programming schedule for the Jointly Operated Facilities;
- (d) System-wide Programs are not intended to duplicate other Programming offered by the Association at the Jointly Operated Facilities, and in cases where the System-wide Programming is similar to Programming offered by the Association, the Association's Programming may remain in place, at the option of the Association;
- (e) required qualifications for instructors offering System-wide Programs will be set by the Park Board;
- (f) where possible, the Park Board will provide the Association with sufficient notice of which System-wide Programs are selected for offering at the Jointly Operated Facilities for a particular Programming registration cycle to allow the System-wide Programs to be scheduled and implemented through the normal Programming committee process;
- (g) up to all five System-wide Programs may be offered during each Programming registration cycle/season; and
- (h) the parties agree that the Park Board may determine it is more appropriate to offer some System-wide Programs at a community centre that serves a region of the City of Vancouver (eg. East/West, North/South), such as in a hub-based model; and
- (i) the offering of System-wide Programs to the public will not commence prior to the beginning of the second year of the Term.

The Park Board acknowledges that the Association will not be required to incur additional costs to fund the System-wide Program and the Park Board agrees that, on an annual basis, it will reimburse the Association for any out-of-pocket costs incurred by the Association to run the System-wide Programs which are not offset by revenue from the System-wide Programs. The Park Board will reimburse the Association within a reasonable period of time after receipt of financial statements or records that detail the annual revenues, expenses and losses resulting from the System-wide Programs. If the Association incurs up-front costs for equipment or supplies that are required to offer a System-wide Program, the Park Board will reimburse the

Association within a reasonable period of time after receipt of the invoices detailing the outof-pocket costs incurred by the Association for equipment or supplies. If the Park Board determines that a System-wide Program should be offered free of charge, the costs to run the System-wide Program will be paid for by the Park Board, and the Park Board may use funds collected through the Operations Fee for such purpose.

The Park Board will undertake a regular review of outcomes associated with System-wide Programs in collaboration with the CCAs. The Park Board may decide to replace a System-wide Program from time to time, including if such program underperforms or becomes redundant.

7.4 Quality Control and Evaluation

The parties will work together to implement a system-wide quality control mechanism for Programming and the Association will review Programming using the results from the quality-control system and on an as-needed basis.

The Park Board and Association will meet to review System-wide Programs at regular intervals.

8. ASSOCIATION MEMBERSHIP

The Park Board acknowledges that the Association, as a registered society, requires active membership for governance purposes and to ensure active involvement of the community in the Association; however, the parties agree that membership in a CCA is not a pre-requisite and will not be required for the public to access, register for or participate in any Programming or services provided by the Association at or from the Jointly Operated Facilities. All users may register for any Programming or other Association service offered at or from the Jointly Operated Facilities without being a member of a CCA.

At the election of the Association, membership in the Association will be offered to patrons of the Jointly Operated Facilities when registering for Programming or services. If memberships are offered on registration, they may be opt-in or opt-out, at the discretion of the Association, and all such memberships shall be free of charge. At the Association's option, the Park Board will load memberships on to a user's OneCard.

The Association agrees that membership in the Association will not provide discounts or priority registration for room rental, Programming or other uses of the Jointly Operated Facilities.

The Association will have direct access to membership and user data information for the Jointly Operated Facilities (including Programming registration information and other data the Park Board collects through the Registration System regarding users of the Jointly Operated Facilities or Satellite Locations) for the purposes of marketing, planning, Association Board governance (including voting) and grant applications. The use and distribution of all such data by the Association must be in compliance with Applicable Laws (and specifically, all privacy and anti-spam legislation).

9. USE ALLOCATION FOR JOINTLY OPERATED FACILITIES

The Association will have responsibility for allocating the use of Jointly Operated Facilities for the purpose of providing Programming and services. The Association agrees that, at the time of booking, priority in the allocation of space in the Jointly Operated Facilities will be given to:

(00704256v1) 17

- (a) Association offered Programming and services as a first priority;
- (b) use by not-for-profit community organizations for the purposes of physical, recreational, cultural, educational, social and capacity-building programs as a second priority;
- (c) use by other not-for-profit community organizations for purposes other than recreation, culture or arts as a third priority; and
- (d) private rentals as a fourth priority.

Notwithstanding the above priority allocation, from time to time, the Park Board may require first priority short term, reasonable use of space within the Jointly Operated Facilities for Park Board or City civic priorities such as special events (e.g., municipal, provincial and federal elections, major sporting or cultural events) or short term use for urgent or emergency purposes. The use of space in the Jointly Operated Facilities by the Park Board or the City will be free of charge, but the Park Board or City will be required to pay for additional out-of-pocket expenses incurred by the Association in respect of such usage.

The Park Board will provide the Association with as much notice as reasonably possible in the circumstances of any such use of space by the Park Board or the City within the Jointly Operated Facilities. The Park Bard will work diligently to mitigate any negative impacts to users of the Jointly Operated Facilities resulting from such use. Where possible, the Park Board will assist the Association to locate alternative space to accommodate any Programming or events that may be displaced as a result of the Park Board or City's use. For greater certainty, the Park Board and City will not be required to compensate the Association for lost opportunity or revenue as a result of such usage, unless agreed to by the parties in advance.

Subject to long-term agreements for periods of one year or greater, the Association will have control of room rentals in the Jointly Operated Facilities and will set pricing for such rentals. Any long-term agreement (for periods of one year or greater) for use of space in the Jointly Operated Facilities must be approved in writing and in advance by the Park Board. Any long-term agreements approved after the Effective Date will be noted in Appendix C, as it is updated from time to time.

The Association will have access to reasonable office space in the Jointly Operated Facilities for administrative use, including storage space, which may be used for storing the Association's files and equipment. Requests by the Association for new or additional office or storage space will be subject to the availability of suitable space.

10. CHILDCARE

The Association will ensure that all Licensed Childcare operated in the Jointly Operated Facilities will meet the standards set by the Park Board and City and will be in compliance with any other relevant standards and Applicable Laws, including the *Community Care and Assisted Living Act* (BC), as it may be amended or replaced from time to time.

11. AFFILIATED GROUPS

In accordance with the Association's responsibility for allocating the usage of space in the Jointly Operated Facilities, the Association may, at its discretion, enter into arrangements for

use of space with other community groups or clubs focused on recreation, culture or arts and which wish to operate out of the Jointly Operated Facilities. The Association acknowledges and agrees that it may not grant rights to use space that is located outside of the Jointly Operated Facilities or commit to any arrangement that places an obligation on the Park Board, without the prior written consent of the Park Board. All requests from a club or group that involve use of space outside of the Jointly Operated Facilities or which may place any obligation on the Park Board (for example, indemnity) must be brought to the Park Board for consideration. Any arrangements entered into by the Association with an affiliated group are subject to the terms and conditions of this Agreement and may not contradict or conflict with this Agreement.

12. HUMAN RESOURCES

12.1 Association Personnel and Volunteers

The Association is and shall continue to be the employer or contracting party, as applicable, of all personnel retained by the Association to provide services on behalf of the Association, including for Programming delivery (collectively, the "Association Personnel"). The Association is and will continue to be responsible for all associated employer and contractor related liabilities and legal obligations regarding Association Personnel. The Association will retain final authority over matters related to employment or service contracts with Association Personnel, including conditions of work, remuneration, evaluation, direction and termination.

The Association will be responsible for all volunteers engaged to support Association activities, and will be responsible for all related liabilities and obligations, including appropriate screening and oversight (e.g., obtaining criminal record checks, checking references, and interviewing potential volunteers), unless formally delegated to the Park Board in accordance with Section 12.2.

The Association acknowledges and agrees that the City and Park Board will not indemnify the Association for liabilities or Breaches arising in relation to the Association's relationships and obligations to Association Personnel or Association volunteers, including statutory and regulatory liabilities. The Association agrees to obtain, maintain and pay for the insurance coverages required under Section 14.6 of this Agreement.

The Association hereby agrees to indemnify the City and the Park Board for:

- (a) Breaches relating to employment or service contracts with Association Personnel or Association volunteers (for example, improper termination or human rights complaints); or
- (b) non-compliance with any statutory or regulatory obligations that Association has in respect of Association Personnel or volunteers (for example, completing and remitting statutory deductions or completing criminal record checks),

unless such obligations are delegated by the Association to the Park Board and accepted by the Park Board in accordance with Section 12.2, or are otherwise caused by the negligence or willful misconduct of the Park Board or Park Board Personnel.

12.2 Delegation of Responsibility to Park Board

Subject to the prior agreement of Park Board, the Association may delegate responsibilities or functions related to the engagement or management of Association Personnel or volunteers to Park Board Personnel. The responsibilities and functions being delegated must be detailed in writing and confirmed by the parties in writing.

To the extent that such obligations are properly delegated to and accepted by the Park Board in accordance with this Section 12.2, or are caused by the negligence or willful misconduct of the Park Board or Park Board Personnel, then the City, through the Park Board, hereby agrees to indemnify the Association for:

- (a) Breaches relating to employment or service contracts with Association Personnel or Association volunteers;
- (b) Non-compliance with any statutory or regulatory obligations with respect to Association Personnel; or
- (c) Breaches that occur as a result of the Park Board failing to properly carry out oversight obligations related to Association Personnel or Association volunteers.

12.3 Park Board Personnel

The Park Board/City will employ and pay for Park Board Personnel required and assigned to operate and maintain the Jointly Operated Facilities, subject to the Association's obligations to pay the Staffing Cost Recovery Payment. The Association and the Park Board will develop a process to review staffing levels at the Jointly Operated Facilities to consider the unique requirements of the Jointly Operated Facilities. The final authority for staffing decisions resides with the Park Board. The Park Board and CCAs will also review staffing across the Community Centre Network on a system-wide basis to identify any gaps or inequities in staffing and consider options to address identified issues.

The Association may provide an orientation to new regular full time, regular part time, and temporary full time Park Board Personnel with information on Association operations, policies and roles.

The Park Board will notify the Association of staff vacancies for Park Board employed regular full time, regular part time, and temporary full time positions. In the event of staff vacancies, the Park Board will develop a plan for interim coverage and permanent recruitment and share the plan with the Association.

The Association acknowledges that the Park Board and Park Board Personnel are subject to collective agreements and individual employment contracts, City policies, and applicable legislation. The Park Board will share the relevant collective agreements and policies with the Association.

12.4 Role of Recreation Supervisor

The Recreation Supervisor is Park Board Personnel and is accountable to the Park Board. The Recreation Supervisor is the management representative of the Park Board at the Jointly Operated Facilities and the local contact for the Association with respect to matters under this

Agreement. Among other responsibilities as may be determined by the Park Board from time to time, the role of the Recreation Supervisor is to:

- provide official oversight for the Jointly Operated Facilities;
- provide support to the Association Board to assist with the delivery of Programming and services offered at or through the Jointly Operated Facilities;
- on request, assist the Association by providing support on board recruitment, director retention, training and governance;
- work with the Association on the planning, delivery, evaluation and administration of all Programming and Association services offered at or through the Jointly Operated Facilities;
- maintain effective communication with the Association President and Association Board, and regularly attend Association Board meetings (excluding in camera meetings); and
- perform other duties which may be necessary to address the unique needs of the Jointly Operated Facilities, provided such additional duties are approved by the Director of Recreation.

The Recreation Supervisor will work cooperatively with and on behalf of the Association to support the Association with delivery of its services and Programming at or through the Jointly Operated Facilities. For requests which may impact unionized staff or operations of the Jointly Operated Facilities (for example, concerns regarding hours or duties of staff), the Association must give reasonable advanced notice to the Recreation Supervisor to consider any such request.

12.5 Hiring Decisions

The Association will have Input into the selection of key Park Board Personnel at the Jointly Operated Facilities, including the community centre recreation programmer, youth worker and new hires to fill the Recreation Supervisor position. Input will also include the opportunity to participate in interviews with shortlisted candidates and to make recommendations for consideration by the Park Board. The Association may provide a description of the unique attributes, needs and nature of the Jointly Operated Facilities and the neighbourhood being served to accompany the job description and posting for a Recreation Supervisor position. For lateral transfers of a Recreation Supervisor, the Park Board will engage in discussions with the Association and consider the views of the Association on the proposed transfer.

The Park Board will take into account the unique attributes, needs and nature of the Jointly Operated Facilities and the neighbourhood being served, when considering the qualifications, knowledge and skills of the Recreation Supervisor selected for the Jointly Operated Facilities. The full and final authority on staffing decisions (including hiring selection and work assignments) will rest with the Park Board.

12.6 Feedback and Issue Management

The Association and the Park Board will each ensure its respective Personnel and volunteers conduct themselves in a professional and respectful manner and in compliance with all Applicable Laws.

The Park Board and the Association agree that communications between the parties and Input given by one party to the other regarding day-to-day operations at the Jointly Operated Facilities will be constructive, respectful and aimed at supporting the best provision of services to the community.

The Park Board will provide the Association with the opportunity to provide feedback on the performance of the Recreation Supervisor when deemed necessary by the Association and on an annual basis. The Association may also share feedback on other Park Board Personnel (other than the Recreation Supervisor) with the Recreation Supervisor at any time.

If either party has concerns with the conduct of the other party's respective Personnel and volunteers, they may bring issues to the attention of the Recreation Supervisor or the Association President, as applicable, at any time. Unresolved matters may be escalated by either party in accordance with the dispute resolution process outlined in Section 18.

13. RENOVATIONS AND MAINTENANCE

13.1 Renovations, Upgrades and Maintenance of Jointly Operated Facilities

The Association acknowledges that its purpose is not to carry out upgrades (including renovations), maintenance or repair to the Jointly Operated Facilities or Common Spaces, and that all upgrades, maintenance or repair are the responsibility of and will be carried out by the Park Board, unless the Park Board, in its sole discretion, provides the Association with express written approval to carry out work. All approvals for work must be requested in advance and will be considered on a project by project basis.

The Park Board, through the City, has conducted building condition assessments for assets under the control of the Park Board, including the Entire Facility, which assessment reports are updated periodically and on a scheduled basis. Maintenance, repair, replacement and upgrade plans developed for buildings, including the Entire Facility, are based on the outcome of the building condition assessments, service delivery needs, priorities for health and safety, state of repair of the buildings and availability of resources.

The Association may provide suggestions to the Recreation Supervisor, with a copy to the Director of Recreation, for consideration by the Park Board regarding what maintenance, repair or upgrades may be the most beneficial to the Jointly Operated Facilities and community and to suggest priorities for maintenance, repair or upgrades to the Jointly Operated Facility or Common Spaces. The Park Board will use reasonable efforts to engage the Association as a stakeholder when making decisions about priorities for maintenance, repair or upgrades to the Jointly Operated Facility or Common Spaces. The parties will agree on a more detailed process through which the Association may provide suggestions.

When determining the annual priorities for maintenance, repair and upgrades of the Jointly Operated Facilities and Common Spaces, the Park Board will consider any suggestions provided by the Association to the Recreation Supervisor. The Association acknowledges that the Park

Board makes decisions regarding priorities for repair, maintenance and building upgrades on a system-wide basis considering, without limitation, health and safety, state of repair and the availability of resources, and the final decision with respect to all maintenance, repair and upgrade work for the Jointly Operated Facility and Common Spaces will rest with the Park Board.

The Park Board will share with the Association the list of annual priorities and anticipated maintenance, repair and upgrade work to be carried out for the Jointly Operated Facilities and Common Spaces. The Park Board will work with the Association to determine appropriate scheduling of the planned work and to manage any impacts of the work in order to minimize disruption to the public.

If the Association wishes to promote projects, enhancements or improvements to the Jointly Operated Facilities, Common Spaces or Outdoor Areas which are not within the scope of the annual repair, maintenance and upgrade plans for the Entire Facility, the Park Board will work with the Association to facilitate the optional enhancement projects, considering availability of resources and system-wide priorities; however, the Association acknowledges that the final decision regarding such optional projects or enhancements rests with the Park Board. If the Park Board approves of the optional project or enhancement, the Park Board will use reasonable efforts to facilitate any such project, and if the Association is concerned that the Park Board is not using reasonable efforts, then they may escalate the matter to the Director of Recreation. If the Association is still unsatisfied after discussions with the Director of Recreation and both parties using reasonable efforts to resolve the matter at that level, the Association may further escalate the matter to the General Manager of the Park Board and subsequently, to the Park Board Commissioners.

If standards for Licensed Childcare change such that maintenance or renovations are required to the Jointly Operated Facilities for existing Licensed Childcare facilities to meet the updated standards, the Association will identify the required maintenance, upgrade or renovation required and provide a list of required work to the Park Board for final review and approval. The parties will share in the cost of the required work and a cost-sharing arrangement shall be negotiated and agreed upon for the required work. Any disagreement on the cost-sharing agreement will be resolved in accordance with Section 18. Once the cost-sharing arrangement is agreed to, the Park Board will be responsible to carry out the work within a reasonable period of time, or such shorter period of time as is required by the licensing authority for the Community Care and Assisted Living Act, SBC 2002 c. 25 to ensure on-going compliance of the Licensed Childcare facility with Applicable Laws.

13.2 Maintenance of Other Assets

With respect to equipment, appliances and other assets used in the Jointly Operated Facilities or Common Spaces that are not expressly identified as the responsibility of the Association or the Park Board elsewhere in this Agreement (such as in Sections 13.3, 14.2(b) and 16.2), those equipment, appliances or other assets (such as kitchen appliances, furniture, Programming equipment etc.) will be maintained and repaired in accordance with the current practices of the parties, and at that parties cost, regardless of whether the particular piece of equipment, appliance or other asset is determined to be Association Equipment or is owned by the Park Board. For all new equipment, appliances and assets purchased after the Effective Date, the parties will agree on maintenance, repair and cost responsibilities in writing before the asset is purchased and installed. In the absence of agreement, the owner of the asset will be responsible to carry out and pay for all maintenance and repair.

13.3 Planning for Surrounding Areas

When planning for parks and recreational amenities in the vicinity of the Jointly Operated Facilities, the Park Board agrees that it will use reasonable efforts to engage with the Association as a stakeholder, where appropriate, and seek its feedback.

13.4 Art Installations and Murals

All art installation and murals proposed for the Jointly Operated Facilities, Common Spaces or Outdoor Areas require the prior written approval of the Park Board. If an art installation or mural is purchased or commissioned by the Association, the Association will be responsible for the upkeep and maintenance (including all costs) of the art installation or mural and the eventual removal of the art installation or mural (including all costs), if and when required by the Park Board. The Association may request, on a case by case basis, a relaxation of these requirements if they cause financial hardship. All relaxations must be requested prior to the purchasing or commissioning of the art installation or mural.

13.5 Vehicles

All vehicles owned by the Park Board or the Association used in connection with the provision of Programming or services from the Jointly Operated Facilities must be maintained by Park Board fleet management and in accordance with Park Board standards. All maintenance and repair costs will be the responsibility of the owner of the vehicle. Any costs that will be responsibility of the Association will be discussed in advance and consented to by the Association before they are incurred.

14. FINANCE

14.1 Revenue

(a) Revenues and Cash Flow

The Association will receive the following revenue: (i) all Facility-Generated Revenue; and (ii) all grant, bequest, fundraising or donation funds directed to the Association.

Unless otherwise agreed to by the parties, all Centrally Processed Revenue will be collected by the Park Board using the Registration System. The sources of revenue included in "Centrally Processed Revenue" may be amended by the parties from time to time and any amendments will be noted in Appendix C, as updated from time to time. Authorized Deductions will be deducted from the Centrally Processed Revenue, and any other revenue that the Association elects to have processed through the Registration System (for example, donation revenue), before it is paid to the Association. The Park Board will pay all Centrally Processed Revenue, less Authorized Deductions, to the Association on a bi-weekly basis, unless otherwise agreed to by the parties. Authorized Deductions will be processed on the same schedule as the payment schedule. The Park Board will provide the Association with a bi-weekly payment report which will include an itemized list of all deductions. The Park Board will not withhold payment of any Centrally Processed Revenue collected through the Registration System, other than for Authorized Deductions, without the Association's prior written consent.

(b) Vending Machine Revenue

The Association acknowledges that the Park Board may enter into vending and supplier contracts ("Vending Contracts") for vending machines or food and beverage distribution rights that apply to the Community Centre Network (such as the current contract between the Park Board and the Coca-Cola Bottling Company dated December 1, 2006). The Park Board agrees that any revenue from vending machines located in the Jointly Operated Facilities which are under a system-wide Vending Contract and received by the Park Board may be shared by the Association and the Park Board at a percentage split agreed upon by the parties. The Association further acknowledges and agrees that Vending Contracts may contain exclusivity clauses which prohibit the sale of competitor's products in the Jointly Operated Facilities and Common Spaces. If the Park Board enters into a Vending Contract which contains an exclusivity clause, the Park Board will share the exclusivity clause with the Association. The Association agrees it shall abide by the terms of any exclusivity clause in the Vending Contract, and shall require its Personnel or those with whom it contracts to do the same. The Association is also permitted to independently operate vending machines in the Jointly Operated Facilities provided they do not violate the terms of any such exclusivity clause. Revenue from any vending machines in the Jointly Operated Facilities that are under contract with the Association (and not part of Vending Contracts) shall be for the benefit of the Association. Subject to the following, vending machines located outside the Jointly Operated Facilities are solely controlled by the Park Board and revenue shall be for the benefit of the Park Board. Any vending in Common Spaces will be considered on a case by case basis.

Any additional financial issues specific to the Association or the Jointly Operated Facilities are addressed in Appendix C to this Agreement.

(c) Use of Revenue

All Facility-Generated Revenue or other revenue generated through the use of the Jointly Operated Facilities, Common Spaces or with the assistance of the Park Board or Park Board Personnel that is received by the Association will be used by the Association to directly benefit the public through the provision of Programming and services at or from the Jointly Operated Facilities, the acquisition of equipment, minor capital and supplies used for the provision of such Programming and services, or for the payment of expenses incurred in respect of the Association's operations pursuant to this Agreement or otherwise in furtherance of the Association's purposes as set out in its constitution, unless otherwise agreed to by the Park Board.

The Park Board also acknowledges that the Association may use reasonable and appropriate amounts of Facility-Generated Revenue to cover necessary expenses (for example, rent payments or janitorial costs) and minor maintenance at the existing Satellite Locations used for Programming.

14.2 Expenses

(a) Park Board Expenses

Except for expenses that are the responsibility of the Association under this Agreement, the Park Board is responsible for all costs and expenses associated with the operation, maintenance, repair and any upgrade of the Jointly Operated Facilities and Common Spaces, including the following costs and expenses:

- (i) the wages, salaries and benefits of Park Board Personnel required and assigned to operate and maintain the Jointly Operated Facilities;
- (ii) assessment, upgrade, operation, maintenance and repair of the Jointly Operated Facilities building, Common Spaces and Outdoor Areas and related building systems (such as mechanical, electrical, structural), including the building envelope and core infrastructure (such as roof, walls, foundation, paint and standard flooring and lighting), subject to funding allocated and approved by the Park Board;
- (iii) subject to arrangements made in accordance with Section 13.2, the upkeep, maintenance, repair and replacement of equipment in the Jointly Operated Facilities and Common Spaces that is not Association Equipment;
- (iv) utilities, waste/recycling removal, access control systems, security and janitorial services for the Jointly Operated Facilities and Common Spaces; and
- (v) all other equipment required to operate the Jointly Operated Facilities, such as computers and point-of-sale equipment.

(b) Association Expenses

The Association shall be responsible for the following costs and expenses, except to the extent that the Park Board agrees to be responsible for any such costs in writing:

- (i) all costs and expenses incurred to operate or administer the Association;
- (ii) all costs and expenses incurred to purchase, maintain, operate and replace computers and equipment purchased by the Association for its own use;
- (iii) all costs and expenses incurred to maintain and repair any Association-owned vehicles;
- (iv) all costs and expenses associated with the provision of Programming or Association services offered at or from the Jointly Operated Facilities, including without limitation, expenses associated with Association Personnel, and equipment, supplies and promotional materials for Programming;
- (v) all costs and expenses incurred to purchase, maintain, repair or replace any specialized equipment (such as ballet bars, fitness centre flooring or mirrors) used to offer Programming, to equip and operate Association run fitness centres or Licensed Childcare operated by the Association or under a rental agreement with the Association, or for any other service for which the Association receives the revenue, and provided always that all work involving repairs or alterations must be in accordance with Section 13.1;
- (vi) the purchase, maintenance, repair and replacement of minor capital equipment (such as pottery kilns and woodworking shops) used for Programming or other Association operations, unless the parties agree in writing that the Park Board will pay for or maintain any such equipment in accordance with Section 13.2, and provided always that all work involving repairs or alterations must be in accordance with Section 13.1;

- (vii) subject to arrangements made in accordance with Sections 13.2 or 13.3, the supply, upkeep, maintenance and replacement of Association Equipment, including any art installations or murals purchased or commissioned by the Association;
- (viii) all costs and expenses in respect of any Satellite Locations, including any costs of renting, leasing, maintaining or repairing any Satellite Location;
- (ix) the Staffing Cost Recovery Payments, in accordance with Section 14.2(c);
- (x) costs to maintain the insurance required under Section 14.6;
- (xi) the Authorized Deductions; and
- (xii) any other costs incurred by the Association that are not in respect of the Association's roles and responsibilities under this Agreement.

The Park Board may estimate components of Authorized Deductions if they do not align with the payment schedule agreed upon by the parties, with a reconciliation of Authorized Deductions to be provided by the Park Board to the Association at a minimum of once annually. The Park Board shall provide notice to the Association for any anticipated or actual increases to the Authorized Deductions with as much notice as reasonably possible in the circumstances.

(c) Staffing Cost Recovery Payment

The Association acknowledges and agrees that it is responsible for the payment of the Staffing Cost Recovery Payment.

The Park Board shall obtain the Association's approval for the Staffing Cost Recovery Payment costs and the positions to be covered on an annual basis in alignment with the Park Board annual budgeting cycle, including any salary or wage increases negotiated under the collective agreement annually, and the Association will advise the Park Board of its decision to approve or not approve the costs within 60 days of such request for approval from the Park Board. The Association acknowledges that, if it chooses to not approve all or any portion of the Staffing Cost Recovery Payment costs, the Park Board has the right to adjust staffing levels to reflect the lack of funding, up to and including elimination of the position that was formerly covered by the Staffing Cost Recovery Payment.

Once approved by the Association, the Association acknowledges it is responsible to pay the full actual costs of the Staffing Cost Recovery Payment for the full year, or portion thereof, as applicable. The Park Board will invoice the Association for the costs on a quarterly basis, unless otherwise agreed to by the parties, and the Association agrees to pay the Park Board the amount of the Staffing Cost Recovery Payment in full within 30 days of delivery of the invoice by the Park Board. Association challenges of the amounts billed for Staffing Cost Recovery Payments will be handled in accordance with the dispute resolution process set out in Section 18.

14.3 Budgets and Records

(a) Operating Budget

The Association will, on an annual basis, prior to the end of a Fiscal Year, develop all budgets required for the Association's operations, roles and responsibilities at or from the Jointly

Operated Facilities for the Association's upcoming Fiscal Year, including its estimated Operations Fee payment. The Association will share its budgets with the Park Board prior to the beginning of the Association's upcoming Fiscal Year. If requested by the Association and agreed to by the Park Board, Park Board staff and/or the Recreation Supervisor will support the Association in the creation of these budgets.

The Park Board will, on an annual basis, develop an operating and capital budget for the Community Centre Network. The Park Board's budget shall be subject to approval of the Park Board Commissioners. Park Board staff will review the operating budget of each facility in the Community Centre Network and, in its sole discretion, reserves the right to make adjustments to increase or decrease the operating budget for all or any part of the Community Centre Network, including by increasing or decreasing the budget for the Jointly Operated Facilities.

The Park Board, to the best of its ability, will provide a copy of the annual operating budget for the Jointly Operated Facilities to the Association. The list of maintenance, repair and upgrade items prioritized for the Jointly Operated Facilities and Common Spaces for that given year's annual budget in accordance with Section 13.1 will be included. The Association acknowledges that some amounts in the Park Board budget are centrally budgeted for and will not be broken out on a facility by facility basis.

The Association acknowledges and agrees that the Park Board budget, including allocations to operating and capital resources for the Jointly Operated Facilities, Common Spaces and the Community Centre Network, are in the sole discretion of the Park Board.

(b) Capital Budgets

The Park Board will develop, for recommendation to City Council, annual capital budgets and future forecasting plans, such as longer term capital plans and capital outlook plans for the Community Centre Network or specific to the Jointly Operated Facilities. The Park Board will seek and consider Input from the Association when determining priorities for capital investment in the Jointly Operated Facilities.

The Park Board and the CCAs shall discuss priorities for capital investment in the Community Centre Network, including renewals and replacements of specific community centre facilities, in the System-Wide Planning Sessions, or in such other meetings as the parties may agree to.

The Association acknowledges and agrees that all final decisions regarding capital budgets and expenses are in the sole discretion of the Park Board.

(c) Record Keeping

The Association will establish and maintain books of account relating to its activities at or from the Jointly Operated Facilities (including at Satellite Locations or as otherwise contemplated by this Agreement) in accordance with sound financial policies and procedures.

If the Association conducts activities that are outside the scope of this Agreement, the Association will either: (1) maintain separate records regarding the other income, expenses, operations or business of the Association, or (2) ensure all financial statements and books of account include notes to separately identify all revenue and expenses relating to the Association's activities pursuant to this Agreement.

{00704256v1} 28

(d) Financial Statements

The Association will provide audited financial statements for each Fiscal Year within the earlier of: (i) one week after the audited Financial Statements have been presented to the membership of the Association at its Annual General Meeting, or (ii) within 180 days of the completion of the Association's Fiscal Year. The Association will post audited financial statements in a prominent location on the Association's website. The Association will also publicly report on Facility-Generated Revenue, grants, donations, bequests, and fundraising revenue, savings, statement of Retained Earnings, expenditures, expenditures of Retained Earnings, transfers to foundations or other organizations and revenue generated in respect of Licensed Childcare that is processed outside of the Registration System, either by including these items in the audited financial statements or by preparing a separate report to be posted on the Association's website together with the audited financial statements.

Within 30 days of a request by the Park Board, the Association shall provide to the Park Board an accounting of the Association's income, expenditures and account balances for the last Fiscal Year, or such other period of time as specified in the request.

Notwithstanding the foregoing, upon request at any time and with reasonable notice, the Park Board may audit the Association's books, financial records and accounts regarding operations at or from the Jointly Operated Facilities or expenditures permitted under this Agreement.

(e) Retained Earnings

Prior to the end of the first year of the Term, the Association will develop a plan to spend and/or keep as operating contingency, at the Association's discretion, its Retained Earnings and will update that plan on an annual basis to reflect amounts actually spent in the past year, use of funds, the current balance of Retained Earnings and any revised plans for future spending of the Retained Earnings. The Association will make its plan publically available and will post its plan and all updates on the Association's website. The Association will use its best efforts to execute on its plan and shall make expenditures of Retained Earnings in accordance with the plan, as it is updated from time to time. Subject to the foregoing, the Association agrees that Retained Earnings may only be spent on physical improvements to or capital projects within the Community Centre Network or for the direct provision of Programming or services to the public at or from the Jointly Operated Facilities.

The Association acknowledges and agrees that the buildings and Fixtures comprising the Entire Facility are and will continue to be owned by the City and Park Board, notwithstanding any contribution of funds (including Retained Earnings and grant revenue) by the Association.

14.4 Grants

The Association may, at their discretion, apply for grants from third parties or the City for initiatives. If the grant application is in respect of an activity or program that will be conducted at or from the Jointly Operated Facilities, the grant application, administration, and oversight must be done in collaboration with the Recreation Supervisor and with support of Park Board staff. All new programs and/or capital projects at or through the Jointly Operated Facilities, including those funded by grants, require the prior review and written approval of the Park Board for indemnity purposes pursuant to the Indemnity Agreement.

Where required, Park Board staff will receive training in evaluation methods and grant writing in order to provide support to the Association for grant applications.

If the Park Board receives information about grant programs offered by other government bodies, it will provide that information to the Association to support the Association's initiatives.

All grant revenue will be paid to the Association without any deduction by the Park Board or as otherwise expressly agreed to by the parties in advance, unless grant revenue is processed through the Registration System, in which case the grant revenue will be subject to Authorized Deductions.

14.5 Operations Fee

(a) Payment of Operation Fee

The Association will make an annual Operations Fee payment to the Park Board in an amount equivalent to a percentage of the Association's gross prior year Facility-Generated Revenue and any gross revenue generated in respect of Licensed Childcare that is processed outside of the Registration System. These payments are not a gift and are being made as a commitment by the Association to the principle of equity in the Community Centre Network and as partial consideration for the rights being granted by the Park Board to the Association under this Agreement. The Park Board is considered a qualified donee for the purposes of the Income Tax Act as an administrative arm of the City of Vancouver. Notwithstanding the foregoing, the Association will not be required to make any Operations Fee payment in respect of revenue generated from Programming and services at Satellite Locations offered independently by the Association without the involvement or assistance of the Park Board or Park Board Personnel.

The Association will pay its annual payment to the Park Board within 3 months of the end of the Association's Fiscal Year. The percentage applicable for each year of the Term shall be as set out in the following table:

Year of Term	Percentage of Gross Revenue for Payment
1	0%
2	1%
3-10	2% each year
Renewal Term	2%, or such greater amount as may be approved by the Park Board Commissioners in a public process.

The purpose of the Operations Fee payment is to facilitate reinvestment in the Community Centre Network to achieve the Public Policy goals of the Park Board and/or increase equity across the Community Centre Network. Operations Fee payments will not be used to finance capital projects or provide supplemental funds to the budget of the Park Board.

In situations of financial hardship, the Association may request that the Park Board permit a partial or total exemption from that year's Operations Fee payment. The Park Board will have the right to review the Association's books of account and other related financial information to assess claims of financial hardship. For the purposes of this Section, "financial hardship" shall mean when an Association has encountered unforeseen circumstances which have strained the Association's financial capacity such that the Operations Fee payment will:

{00704256v1} 30

- (i) result in the Association being unable to pay its normal and reasonable operating expenses and compromise the ability of the Association to provide Programming and services at the Jointly Operated Facilities; or
- (ii) otherwise reasonably be seen to create a threat to the viability of the Association.

(b) Allocation of Operations Fee

The Operations Fee payments will be held by the Park Board and the funds will be allocated exclusively to achieve Public Policy goals within the Community Centre Network, including by enhancing access to, quantity and/or quality of Programming and services delivered throughout the Community Centre Network or to increase equity across the Community Centre Network.

The Park Board will consider any Input provided by the CCAs regarding the use of the Operations Fee funds, but in all instances the Park Board will retain decision-making authority.

The Park Board will prepare an annual report describing the use and allocation of the Operations Fee funds and related outcomes, and will make the report public.

14.6 Insurance

The Association, at its sole cost and expense, will ensure all Association directors and officers maintain director and officers' liability insurance, including coverage for employer liability where available, and shall provide proof to the City or Park Board upon request.

The Association agrees to obtain, maintain and pay for, for the entire duration of the Term, Employment Practices Liability coverage, with the City and Park Board as additional named insureds, and provide evidence of this coverage to the Park Board. Insurance coverage must be appropriate to the circumstances and acceptable to the Park Board and/or City, acting reasonably. If the requirements for insurance coverage change during the Term, the City and/or Park Board will provide reasonable notice to the Association of the amended insurance coverage requirements, and the Association shall ensure its coverage meets the amended requirements, as applicable.

14.7 Miscellaneous Operations

Subject to any written agreement between the parties, any and all parking revenue collected at the Entire Facility shall be for the sole benefit of the Park Board.

The revenue from any new commercial operation or concession at the Jointly Operated Facilities shall be allocated to the Park Board; however, the Park Board will discuss the new arrangement with the Association to consider a different allocation of revenue on a case by case basis. The revenue allocation for all commercial operations and concessions existing as of the Effective Date shall be in accordance with the practices detailed in Appendix C.

15. COMMUNICATIONS AND PLANNING

15.1 Communications Between Parties

All communications between the parties shall be respectful and constructive and will be in line with the parties' acknowledgement and respect of their distinct roles and shared objectives.

The Park Board and Association shall maintain regular communications regarding the operation of the Jointly Operated Facilities, provision of Programming and the roles and responsibilities of each party under this Agreement, including through regular meetings between Park Board staff and the Association. In addition to regular meetings, the parties will participate in the System-wide Planning Sessions described below.

The Park Board will advise the Association in a reasonable timeframe of any plans, including building repair or maintenance projects, which may impact the use of or access to the Jointly Operated Facilities. Except in the case of an emergency, the Park Board will provide reasonable advance notice in writing to the Association when such a disruption is likely. The Park Board will use reasonable efforts to minimize the adverse effects of any such plans.

15.2 System-wide Planning and Communication Sessions

To facilitate system-wide planning, communication and outcomes measurement across the Community Centre Network the Park Board and representation from all CCAs will meet a minimum of twice annually (each, a "System-wide Planning Session"). The System-wide Planning Sessions will be scheduled by the Park Board and the Park Board will use reasonable efforts to select dates that work for the greatest number of CCA representatives. The Park Board will give a minimum of 30 days' notice of a date that is selected for a planning session. If a director of a CCA is not able to attend, they will select a suitable delegate to attend the applicable System-wide Planning Session.

The System-wide Planning Sessions will be used to review or develop proposed Park Board policies that affect the Community Centre Network, share research and demographic information conducted by any of the parties that is relevant to the Community Centre Network, review or conduct capital planning, hold governance training sessions, discuss and consider any desired changes to this Agreement prior to the Renewal Term, and address any other topics that are of interest to the CCAs or the Park Board. Topics for discussion may be proposed by the Park Board or any of the CCAs. The System-wide Planning Sessions will be focused on topics that are applicable system-wide to the Community Centre Network. Issues that are specific to a particular facility (e.g., facility maintenance and upgrades, marketing and communication plans, and staff training) are not appropriate for discussion at the System-wide Planning Sessions and will be addressed locally with the respective Recreation Supervisor and respective individual Association, as is currently the practice.

The System-wide Planning Sessions are not intended to replace Park Board participation in other regular meetings with the Association.

System-wide Planning Sessions may be held more than twice annually if agreed to by the Park Board and the majority of the CCAs.

16. INFORMATION AND BUSINESS SYSTEMS, TECHNOLOGY & TELECOMMUNICATIONS

16.1 Registration System

The Association acknowledges and agrees that a common recreation and registration management system which is designated, owned and operated by the City/Park Board will be exclusively used for all administrative functions at or related to the Jointly Operated Facilities and that no other system may be used for those functions. Administrative functions include processing, recording and storing all Programming, registration and rental information,

[00704256v1] 32

preparing instructor contracts, processing and recording all payments for Centrally Processed Revenue, customer account inquiries and providing front desk and customer service. Payment in all forms will be processed through the Registration System. The Registration System may also be used to record Association memberships and to process donations, fundraising revenue, bequests and/or grants at the election of the Association. All financial transaction processed using the Registration System will be subject to Authorized Deductions.

The Registration System will be used to generate reports that detail all funds received, payment information, registration statistics, and provide information for measuring program performance. The reports will be generated on a schedule agreed to by the parties and will be made available to both the Association and the Park Board. The Association will also have direct access to other information from the Registration System relating to Association programs/services and membership. The Park Board will conduct regular reviews of the Registration System for accuracy.

As required, the Park Board will meet with the Association to resolve non-technical issues regarding the Registration System and will provide or seek technical support as required.

At the election of the Association, the Park Board will work with the Association to develop and implement a new process to secure rental and damage deposits so that deposits may be secured with a credit card or cheque on file and not be processed through the Registration System such that transaction fees are avoided for refundable deposits.

The Registration System in use as of the Effective Date is "Activenet", a recreation management software provided by by Active Network Ltd. ("ActiveNet"). The Park Board and the CCAs have agreed to more specific details regarding the use of Activenet, which are detailed in Appendix D to this Agreement.

16.2 Ownership of Systems

All information technology software, business systems, hardware, data, wireless hardware and software, information technology infrastructure, and telecommunications equipment installed or utilized in the Entire Facility with connection to City Networks (collectively, the "IT Assets") will be owned, supplied, managed, maintained and repaired by the City on behalf of the Park Board.

If the Association wishes to install any Association-owned software on City or Park Board computers, it must first obtain the prior written approval of the Park Board and, if such software will connect to the City Network, the prior approval of the City IT department.

The Association may purchase its own computer equipment, provided that the Association equipment is not connected to the City Networks or other internet service, except for City public wifi, or used for any business function associated with the Jointly Operated Facilities, and all uses must comply with the terms of this Agreement. The Association will be responsible for all maintenance, repair and replacement of its own equipment.

16.3 Security Systems

All existing or future surveillance systems, security systems and security monitoring will be the sole responsibility of the Park Board. The Association may not install any surveillance, security or monitoring systems or technology in the Jointly Operated Facilities or Common Spaces.

16.4 Management of Data

Both parties acknowledge the unique nature of their relationship and operations at the Jointly Operated Facilities and commit to the cooperative and proper joint management of data and personal information obtained through the operations contemplated by this Agreement. The parties acknowledge and agree that all data and other personal information collected in the course of their operations at or from the Jointly Operated Facilities or otherwise administered using IT Assets will be managed in compliance with all Applicable Laws and will only be used for the original intent for which it was collected.

16.5 Use of Business Systems and Information Technology

The parties use of the IT Assets, including the Registration System, will be subject to the following conditions:

- (a) Applicable Laws: Usage of all IT Assets must be in compliance with Applicable Laws including, for greater certainty, all privacy legislation and the *Freedom of Information and Protection of Privacy Act* (FOIPPA), RSBC 1996 c. 165, as such laws may be amended or replaced from time to time;
- (b) Security: Usage of all IT Assets will be in accordance with City information technology security policies and standards;
- (c) PCI: All IT Assets will be maintained in compliance with the current Payment Card Industry (PCI) standards, as applicable;
- (d) Access Rights: Network and software access will be granted in accordance with City information technology Identity Management, network access, and applications access policies and standards; and
- (e) Hardware: Usage of information technology hardware will be in accordance with City information technology "acceptable use" policy,

as all such policies and standards may be amended or replaced from time to time.

17. OWNERSHIP OF COMMUNITY CENTRE ASSETS

17.1 Community Centre Facility

The Association acknowledges and agrees that the City is the sole owner of the Entire Facility, including the Jointly Operated Facilities. The City is also the sole owner of all other improvements and Fixtures, unless otherwise agreed to by the parties in writing. Jurisdiction for managing the Entire Facility resides with the Vancouver Park Board. The Association acknowledges and agrees that it has no interest, right or title in the Entire Facility and, except for the rights expressly granted to the Association under this Agreement for the Term, it has no independent license, lease or other right to occupy the Entire Facility, now or in the future, whether written or implied.

The Association further acknowledges and agrees that no licence, lease or other right to occupy all or any portion of the Entire Facility will arise or has arisen by virtue of the Association's occupation or operations at all or any part of the Entire Facility, including the Jointly Operated Facilities, or by the past or future conduct of the parties and that any past or future funds

[00704256v1] 34

contributed by the Association towards the Jointly Operated Facilities or any other infrastructure comprising the Entire Facility, including towards the base building, building additions, capital infrastructure or improvements to the Lands, do not give rise to any interest, right or title in all or any portion of the Entire Facility in favour of the Association. All rights of the Association for the use of or access to the Jointly Operated Facilities and Common Spaces arise solely under this Agreement.

Other than items that are agreed to be Association Equipment, all items that are purchased by the Association for the Jointly Operated Facilities or Common Spaces shall be for the benefit of the Entire Facility and are being contributed as an unconditional and irrevocable donation to the Park Board and the City for the purposes of enhancing the Entire Facility. The Association will remain as the owner of all the Association Equipment, as such inventory is updated from time to time.

17.2 Equipment and Assets

The Park Board acknowledges and agrees that the Association is the owner of personal property other than Fixtures and easily removable equipment that were donated to the Association or purchased 100% by the Association using funds at its disposal or which were directed to the Association, unless the parties have agreed otherwise in writing ("Association Equipment"). The Park Board and the Association will create and maintain an up to date inventory of Association Equipment, and any changes, additions or deletions to the inventory must be agreed to by the Association and the Park Board. Disagreements on the ownership of an item will be resolved using the dispute resolution process set out in Section 18.

Other than Association Equipment, equipment leased by the Association, and any equipment or assets of an affiliated group, 3rd party contractor or arms-length leasing party, as of the Effective Date, the parties agree that all property and equipment located in or at the Entire Facility is owned by the Park Board and/or the City.

18. DISPUTE RESOLUTION

18.1 Dispute Resolution Process

It is the belief of both parties that issues should be ideally resolved constructively at the local community centre level. All disputes between the Association and the Park Board must first be discussed by the Recreation Supervisor and/or the Manager of the Jointly Operated Facilities and Association representatives who will attempt to resolve the dispute in a respectful and constructive manner.

If disputes cannot be resolved at the local level within fifteen (15) business days of the matter first being raised in writing or in person, or such longer period of time as agreed to by the parties, then either party may refer the dispute to the Director of Recreation and the Association representative for further discussion and attempt at a resolution.

If the dispute is not resolved by the Director of Recreation and Association representatives within another fifteen (15) business days of when the issue was raised to the Director of Recreation, or such longer period of time as agreed to by the parties, then either party may refer the dispute to the General Manager of the Park Board for further discussion and the parties will attempt to resolve the dispute. If a dispute is referred to the General Manager, the Association and Park Board management staff must each submit a written summary of the

dispute to the General Manager, including all relevant facts and any applicable policies, prior to the dispute being reviewed.

If discussions between the Park Board and Association do not resolve the dispute within thirty (30) business days of the General Manager's receipt of the written summaries of the matter, or such longer period of time as agreed to by the parties, then:

- A. for matters subject to arbitration, the dispute may be resolved by binding arbitration at the election of either of the parties, pursuant to Section 18.2. If agreed to by the parties, the parties may also decide to refer any arbitral issues to mediation or another form of dispute resolution before pursuing arbitration; and
- B. for matters not subject to arbitration, the dispute may be referred to the Park Board Commissioners for further consideration and a final decision. The Park Board Commissioners will render a decision within 2 months of the dispute being referred and receipt of the written summaries of the matter.

The parties agree that the following matters are not subject to arbitration:

- (a) matters under the sole jurisdiction of the Park Board, including:
 - (i) policies and standards established by the Park Board or the City, including Public Policy, provided such policies and standards do not directly contradict the express terms of this Agreement; and
 - (ii) allocation of Park Board budgets and resources;
- (b) compliance with Applicable Laws and corporate policies; and
- (c) the ownership of the Entire Facility, including the Jointly Operated Facilities.

For greater certainty, the Park Board and the Association agree that the implementation of Public Policy initiatives at the Jointly Operated Facilities will be carried out pursuant to Section 6.2 of this Agreement, and any disputes regarding implementation matters will be subject to arbitration.

18.2 Arbitration

All disputes subject to arbitration under this Agreement, including disputes concerning the interpretation of this Agreement and accounting and payment disputes, may, at the election of either of the parties, be referred to and finally resolved by binding arbitration. Matters to be resolved by arbitration will be subject to arbitration in accordance with the Arbitration Act, R.S.B.C. 1996, c. 55. Unless the parties otherwise agree in writing, the dispute will be referred to and finally resolved by arbitration administered by the British Columbia International Commercial Arbitration Centre, pursuant to its Rules for the conduct of domestic commercial arbitrations. The place of arbitration shall be Vancouver, BC. The arbitration will be conducted before a single arbitrator and shall be final resolution for the matter being referred. The administrative fees for the arbitration and the arbitrator's fees and expenses (collectively, the "Arbitration Fees") will be split equally between the Park Board and Association but the ultimate allocation of Arbitration Fees between the parties will be determined by the arbitrator in apportioning costs between the parties at the conclusion of the arbitration.

19. BRAND AND BRAND PROTECTION

The parties acknowledge that they are cooperating to provide valuable services to the public, and that the reputation and public image of both the Park Board and the Association is important to both parties. To ensure the integrity of this Agreement, the parties agree to respect each other's brand and public image and to communicate about each other in a constructive and respectful way in all public materials or forums.

Programming materials (such as the program brochure for the Jointly Operated Facilities) and all signage at the Jointly Operated Facilities will be co-branded and feature the logos of both the Park Board and the Association.

20. CHANGES TO JOINTLY OPERATED FACILITIES

20.1 Additions to Entire Facility

If additions or renovations adding space are made to the Entire Facility, and the new space is to be used for purposes consistent with the scope and offerings of the existing Programming provided by the Association, then this Agreement will be amended as required to include the new space within the scope of the Jointly Operated Facilities. Notwithstanding the foregoing, if the Association is or has been in sustained breach of this Agreement or if the Park Board holds reasonable concerns regarding the Association's ability to operate the new space in a safe, professional and compliant manner, then the new space will not be automatically included in the scope of Jointly Operated Facilities and the Park Board may elect to operate the new space itself or with the assistance or another partner organization.

If, at any time during the Term, including during any Renewal Term, other major changes are made to the Jointly Operated Facilities or Common Spaces, such as the addition of significant new facilities (such as a café or concession), or additions or renovations adding space that is to be used for purposes that are not consistent with the the scope and offerings of the Association's existing Programming or services, then the parties will negotiate an amendment to this Agreement to clarify the roles and responsibilities of the parties with respect to the new facilities. The final decision with respect to any new facilities or space will rest with the Park Board.

20.2 Reduction in Jointly Operated Facilities

If, during the Term, any space in the Jointly Operated Facilities needs to be closed for reasons of safety (for example, removal of asbestos) or is unused by the Association for the purposes contemplated under this Agreement for a sustained period, then such space may be removed from the scope of the Jointly Operated Facilities, as identified in Appendix A. If a portion of the Jointly Operated Facilities are closed for reasons of safety and are subsequently renovated or repaired such that they are able to be safely used again, then such space will be added back to the scope of the Jointly Operated Facilities.

20.3 Replacement of Jointly Operated Facilities

If, during the Term, the Jointly Operated Facilities are to be closed and directly replaced with another facility, and the new facility is to be used for purposes consistent with the scope and offerings of the existing Programming provided by the Association, then this Agreement will be amended to contemplate any changes to the operations at the new facility and the Term shall

{00704256v1}

continue unaffected. Prior to the closure of the Jointly Operated Facilities, the Park Board and Association will discuss the arrangements for the new or replacement facility and enter into a modification to this Agreement to addresses any changes to the facilities or the operating arrangements. If the Jointly Operated Facilities are closed prior to the replacement facility being opened, then the Term shall be extended for a period of time equivalent to the number of days between the closure of the Jointly Operated Facilities and the opening of the new facility.

Notwithstanding the foregoing, if the Association is or has been in sustained breach of this Agreement or if the Park Board holds reasonable concerns regarding the Association's ability to operate all or any component of the replacement facility space in a safe, professional and compliant manner, then the Park Board may elect to terminate this Agreement and not extend this Agreement to all or any portion of the replacement facility.

20.4 Closure of Jointly Operated Facilities

If, during the Term, the Jointly Operated Facilities are closed then this Agreement will terminate and any remaining term shall expire.

21. TERMINATION OR NON-RENEWAL

21.1 Breach by Association and Park Board's Right to Terminate

In addition to the Park Board's rights of termination or non-renewal set out in Sections 3.2, 20.3 and 20.4, the Park Board may terminate this Agreement with a minimum of 90 days prior written notice in the event of a sustained, material, un-remedied breach of this Agreement by the Association; provided that, the right to terminate will be subject to the following:

- (a) the termination notice must specify the breach in reasonable detail;
- (b) the Association will have a cure period of 60 days to remedy the breach, except in the case of emergency, in which case a shorter cure period may be specified. In the event of emergency, the Park Board may take reasonable action to cure the breach on behalf of the Association at the expiry of the notice period, or sooner as may be specified in the notice of breach;
- (c) the Association must make reasonable and diligent efforts to resolve the breach in a manner satisfactory to the Park Board in a timely manner;
- (d) If the Association does not satisfactorily remedy the breach in the cure period, the notice to terminate will be deemed effective and the Agreement will automatically terminate at the end of the 90 day notice period; provided that, if more than 60 days is required to remedy the breach (using reasonable resources and acting diligently), then the Association will not be considered to be in breach after 60 days (except in the case of emergency) if, immediately after receiving the notice of breach, the Association commenced the remedy of the breach, immediately notifies the Park Board of the steps it will take to remedy the breach, and has proceeded to diligently and continuously remedy the breach so that the breach is cured as soon as reasonably possible; and
- (e) If the Association fails to continuously and diligently work towards resolving the breach, the Park Board may terminate this Agreement with no further notice required.

If the Association disagrees that the action or event specified in the notice constitutes a material breach of the Agreement, the Association may invoke the dispute resolution mechanism. If the alleged breach is of a serious nature, in the reasonable opinion of the Park Board, the Park Board may elect to expedite the dispute resolution mechanism and refer the dispute directly to the Park Board General Manger or arbitration, if applicable.

For the purposes of this Agreement, "material breach" of the Association shall include failure of the Association to comply with Public Policy, breach of Applicable Laws, fraudulent conduct, misappropriation of funds, financial wrongdoing or criminal activity on the part of the Association, an unresolved default under section 4.2, breach of section 14.1(c), sustained failure of the Association to provide financial information required under this Agreement, discrimination or human rights violations on the part of the Association, or any conduct by the Association which is directed against the public interest.

In the event of termination of this Agreement, all outstanding payments between the parties will become immediately due and payable and subject to withholding, and the parties will work cooperatively to transition operations to the Park Board or its designate to ensure minimal disruption to the public.

21.2 Breach by Park Board and Association's Right to Terminate

In addition to the Association's rights of non-renewal set out in Section 3.2, the Association may terminate this Agreement with a minimum of 90 days prior written notice in the event of a sustained breach of this Agreement by the Park Board.

If the Association considers the Park Board to be in breach of this Agreement at any time, the Association must provide a notice to specify the breach in reasonable detail. The Park Board will use reasonable and diligent efforts to resolve any breach of this Agreement in a timely manner. If the Park Board does not satisfactorily remedy the breach within 60 days, the Association may terminate this Agreement at the end of the notice period for termination (if notice of termination was provided) or may refer the matter to dispute resolution for further discussion and attempted resolution. In the event of termination of this Agreement, all outstanding payments between the parties will become immediately due and payable and subject to withholding, and the parties will work cooperatively to transition operations to ensure minimal disruption to the public.

22. NOTICES

Any notice, approval or request required or permitted to be given under this Agreement shall be in writing and may be given by delivering such notice, approval or request to a representative of the party for whom it is intended or by mailing such notice, approval or request by prepaid registered mail from any post office in British Columbia:

(a) to the Park Board:

Board of Parks and Recreation 2099 Beach Avenue Vancouver, British Columbia V6G 1Z4 Email: _____

Attention: General Manager

(b) to the Association:

Killarney Community Centre Society 6260 Killarney Street Vancouver, British Columbia V5S 2X7

Attention: President

or at such other address as the parties may from time to time advise by notice in writing. Any such notice, approval or request shall be deemed to have been received on the date of delivery of such notice, approval or request or, on the third business day next following the date of such mailing if mailed as aforesaid, provided that if mailed should there be, between mailing and the actual receipt of such notice, approval or request, a mail strike, slowdown or other labour dispute which might affect the delivery of such notice, approval or request, such notice, approval or request shall only be effective if actually delivered.

23. TERMINATION OF PRIOR JOA

The Association and the Park Board hereby acknowledge and agree that this Agreement replaces the Prior JOA, and the Prior JOA is hereby terminated with effect on the Effective Date, and shall be of no force and effect from and after the Effective Date.

24. MISCELLANEOUS

24.1 Governing Laws

This Agreement, regardless of where executed or performed, will be governed by and construed in accordance with the laws of the Province of British Columbia.

24.2 Enurement

This Agreement shall enure to the benefit of and be binding upon the Park Board and the Association and their respective successors and permitted assigns.

24.3 Severability

All the obligations and covenants contained in this Agreement are severable, so that if any one or more of the obligations or covenants are held by or declared by a court of competent jurisdiction to be void or unenforceable, the balance of the obligations and covenants shall remain and be binding.

{00704256v1}

24.4 Non-Derogation

Nothing contained or implied in this Agreement will derogate from the obligations of the Association under any other agreement with the City or the Park Board or, prejudice or affect the City or Park Board's rights, powers, duties or obligations in the exercise of its functions pursuant to the *Vancouver Charter* as amended from time to time and the rights, powers, duties and obligations of the City and Park Board under all public and private statutes, by-laws, orders and regulations, which may be as fully and effectively exercised in relation to the Jointly Operated Facilities as if this Agreement had not been executed and delivered by the Association and the Park Board.

24.5 Time of the Essence

Time shall be of the essence of this Agreement.

24.6 Captions

The captions and headings throughout this Agreement are for convenience and reference only and the words and phrases contained therein shall in no way be held or deemed to define, limit, describe, explain, modify, amplify or add to the interpretation, construction or meaning of any provision of or the scope or intent of this Agreement nor in any way affect this Agreement.

24.7 Interpretation

Words importing the singular will include the plural and vice versa, and words importing gender will include the masculine, feminine and neutral genders.

24.8 Waiver

The parties acknowledges and agrees that no failure by either party to exercise and no delay in exercising any right or powers of enforcement under this Agreement by either party shall operate as a waiver thereof. The remedies herein provided shall be cumulative and not exclusive of any other remedies provided by law or equity.

24.9 Entire Agreement

This Agreement and, for greater certainty, each of the Appendices affixed hereto constitute the entire agreement and understanding between the parties hereto with respect to the subject matter hereof, and supersedes all previous communications, representations and agreements, whether oral or written, with respect to the subject matter hereof. All amendments to this Agreement must be made in writing and signed by both parties. For greater certainty, the parties agree that the Appendices to this Agreement may be adjusted by the parties at any time, upon mutual agreement, as practices or operations of the parties may change over time. The parties further acknowledge and agree that the Indemnity Agreement continues to be in full force and effect, subject to the terms and conditions of that agreement, and has not been modified in any way by this Agreement.

24.10 Appendix C Prevails

In the event of any conflict between the terms in the body of this Agreement and the terms of Appendix C attached hereto, the terms of Appendix C will prevail.

24.11 Assignment

The Association shall not assign its rights or obligations hereunder without the prior written consent of the Park Board, which consent may be withheld in the sole discretion of the Park Board.

24.12 Independent Legal Advice

Each of the parties acknowledge that they have been advised to obtain and have been afforded sufficient opportunity to obtain independent legal advice prior to entering into this Agreement.

24.13 Collective Agreements

The Association acknowledges and agrees that nothing in this Agreement supersedes any duties and responsibilities of the Park Board or the City under any and all collective agreements with trade unions certified to represent employees of the Park Board/City.

24.14 City as Public Body

The parties to this Agreement each acknowledge that the City is subject to the provisions of the Freedom of Information and Protection of Privacy Act (British Columbia), as the same may be amended or superseded from time to time ("FOIPPA") and that as a public body under FOIPPA, the City may be requested to disclose confidential information provided to it by the Association and other CCAs, including budgets and records of revenue and expenses. The Association agrees that any such disclosure by the City will not constitute a breach of the City's obligations hereunder.

24.15 Permanent Public Park

Notwithstanding anything contained in this Agreement, if the Jointly Operated Facilities are part of a permanent public park within the meaning of section 490 of the *Vancouver Charter*, S.B.C. 1953, C. 55, as amended, and ceases to be part of such a permanent public park pursuant to section 488 of the *Vancouver Charter*, then this Agreement and the licence granted hereunder will be terminable at the option of the Park Board but all obligations of the Association up to the date of any such termination will survive such termination. Upon termination pursuant to this Section, at the option of the Association and subject to any required approval of Vancouver City Council or the Park Board, the Park Board shall enter into a separate licence agreement with the Association on the same terms and conditions of this Agreement for the remaining balance of the Term.

TO EVIDENCE THIS AGREEMENT the authorized signatories of each of the parties have executed this Agreement to be effective as of the Effective Date.

CITY OF VANCOUVER as represented by its BOARD OF PARKS AND RECREATION by its authorized signatory:	
Signature Signature	Malwim Bromley, General Manager Print Name and Title
KILLARNEY COMMUNITY CENTRE SOCIETY by its authorized signatories:	
WTmply (me 8)	W Tupling Gurst Dresident KCCS Print Name and Title
Signature	AINSCIE KWAN PAST PRESIDENT KCCS Print Name and Title

Approved by resolution of the Board of Parks and Recreation on April 10, 2017.

APPENDIX A

Jointly Operated Facilities

Room Name (if applicable)	Room Number (if applicable)	Location - (i.e. Main Floor, 2 nd Floor, Basement, Other)
Martial Arts	004	Basement
Youth Lounge		Main floor
Main Gym		Main floor
Computer Rm	101	Main floor
Childcare	102	Main floor
Childcare	105	Main floor
Kitchen	104	Main floor
Fitness Centre	106	Main floor
Licensed Childcare Coordinator Office	106A	Main floor
Bookkeeper Office	107A	Main floor
Room 201	201	2nd floor
Board Room	202	2nd floor
Room 203	203	2nd floor
Room 203A	203A	2nd floor
Kitchen	204	2nd floor
Dance Studio	205	2nd floor
Room 211	211	2nd floor
Seniors' Centre	N/A	Separate building adjacent to main Community Centre building
Storage area downstairs	004A	Basement

Common Spaces and the Playground accessed via Rooms 102 and 105 will continue to be used jointly by the parties in accordance with current practices.

The Park Board and the Association acknowledge and agree that, from time to time, Common Spaces such as hallways may be used for registration, ticket sales, informal recreation activity, special events, informal meeting and socialization areas and program participant assembly. Furthermore, these Common Spaces may host art displays, art sales and performance installations, provided they do not interfere with the accessibility, safety and proper functioning of the Entire Facility. Common Spaces frequently host showcases, commemoration plaques and art installations including secure display cases such as award cases and art display cases. Common Spaces can host vending machines, seasonal displays, community bulletin boards, marketing materials and brochure racks. The definition of "Common Spaces" does not preclude that space from being programmed or booked in some circumstances, if appropriate.

APPENDIX B

Access Policy

Policy for public access to the Jointly Operated Facilities is set by the Park Board. The current policy, which may be amended or updated from time to time through the Term of this Agreement at the discretion of the Park Board, is set out below.

OneCard

• OneCards issued from the Jointly Operated Facilities will be co-branded with the Park Board logo and Association logo and all access and use products (i.e. Flexipass, LAP, etc.) will be loaded on to a user's OneCard.

Flexipass

- The Association will accept Flexipasses loaded onto the OneCard at the Association run fitness centre.
- The Park Board will provide a share of Flexipass revenue to the CCA as per the reimbursement rates provided in Appendix 1 attached. The reimbursement rates will be reviewed in the first year of the joint operating agreement.
- The Association may continue to sell their own products (eg. fitness centre cards), but will not require their products to be used.

Leisure Access Program

- The Association will accept the Leisure Access Program ("LAP") passes (for fitness and all Programming) loaded onto OneCards of residents approved by the Park Board.
- The Park Board will accept LAP for swimming, skating and fitness programs and other Park Board services, as determined by the Park Board.
- The current LAP subsidy consists of:
 - o 50% off the daily Park Board drop-in fee for all fitness centres and 50% off the adult Flexipass rate for all fitness centres; and
 - o 50% off the Programming rate for one program per person, per season, per centre so long as a program's minimum registration is met;
- The LAP does not apply to private lessons, special events, or Licensed Childcare.
- The LAP does not preclude other subsidy programs.
- The Park Board agrees to financially compensate the Association for any registration in excess of the current LAP subsidy listed above.

Reimbursement to CCA for Flexipass Use at CCA-Run Fitness Centre

The Park Board collects the revenue when a pass is purchased. Each time the card is used at a CCA-run fitness centre, the below amounts will be reimbursed to the applicable CCA, notwithstanding any discounts that the Park Board may have applied to the purchase of the Flexipass.

ADULT		1 month		12 months
Adult Prices as per Vancouver.ca May 2016	\$46.86	\$45.28	\$120.05	\$382.67
Cost per use assuming pass used every day		\$1.51	\$1.33	\$1.05
Cost per use assuming pass used every other day		\$3.02	\$2.67	\$2.10
Cost per use assuming pass used every other three days		\$4.53	\$4.00	\$3.15
Reimbursement amount per use (average)	\$4.69	\$3.02	\$2.67	\$2.10

YOUTH	10 visit	1 month	3 months	12 months
Youth Prices as per Vancouver.ca (May 2016)	\$33.52	\$31.70	\$84.05	\$267.86
Cost per use assuming pass used every day		\$1.06	\$0.93	\$0.73
Cost per use assuming pass used every other day		\$2.11	\$1.87	\$1.47
Cost per use assuming pass used every other three days		\$3.17	\$2.80	\$ 2 .2 0
Reimbursement amount per use (average)	\$3.35	\$2.11	\$1.87	\$1.47

SENIOR	10 visit	1 month	3 months	12 months
Senior Prices as per Vancouver.ca (May 2016)	\$33.52	\$31.70	\$84.05	\$267.86
Cost per use assuming pass used every day		\$1.06	\$0.93	\$0.73
Cost per use assuming pass used every other day		\$2.11	\$1.87	\$1.47
Cost per use assuming pass used every other three days		\$3.17	\$2.80	\$2 .2 0
Reimbursement amount per use (average)	\$3.35	\$2.11	\$1.87	\$1.47

- The methodology used for calculating the "cost per use" above is as follows:
 - o (1) divide the price of the pass by the number of days (30 days/month, 90 days/3 months, 365 days per year), then
 - o (2) multiply the result by the assumed frequency of use (daily, every other day, every three days).
- The "reimbursement amount per use (average)" is an average of the three cost per use calculations above.
- The reimbursement amounts will increase as the price of the pass increases using the associated calculations as outlined above.

• If there is a new pass offered (i.e., 2 month, 2 years, etc.) the above formula will apply.

CCAs operating a CCA run fitness centre retain the right to continue to sell passes to access their site specific fitness centre. Prices of these passes will continue to be set by the CCA.

Park Board agrees to the following:

- Reimbursement rates for LAC usage are at a 50% discount from the above rates.
- Reimbursement rates and the above transactional fees will apply exclusively to CCArun fitness centres.
- Association-run fitness centre usage passes will not be called Flexipasses.
- Reimbursement and usage reporting is conducted on a quarterly basis by the Park Board.

APPENDIX C

Issues and Financial Matters Specific to Association

- 1. The concession stand located in the ice rink (outside of the Jointly Operated Facilities) is operated by an independent operator under a rental arrangement with the Association. The Association receives the rent paid for the concession stand and is responsible for the maintenance and cleaning of the concession stand and the associated equipment. The Park Board acknowledges that the Association may share these responsibilities with the independent operator, at the discretion and risk of the Association.
- 2. The Association is permitted to rent out to third parties (in accordance with the process set out in Section 9 of the Agreement) the multi-purpose room located in the Park Board operated pool area, which is outside the Jointly Operated Facilities. The Association receives the revenue generated from its rental of the multi-purpose room on the condition that the revenue earned from the rental of the multi-purpose room in the pool is used to purchase equipment and aquatic related program supplies for use in the pool. If there is any excess revenue after all of the necessary or desirable equipment and aquatic related program supplies for the pool have been purchased, that excess revenue will be retained by the Association and may be used for other initiatives which are for the benefit to the community.
- 3. The Association currently funds staffing at the Norquay spray park and Bobolink wading pool and the Association acknowledges that there is a public expectation for these programs to continue. The Association and the Park Board will review the funding for these programs on an annual basis and in advance. The Association agrees that, if it decides to stop funding these programs at approximately the same level as the Association is currently funding, the Association will provide the Park Board with as much notice as possible and will work cooperatively with the Park Board to minimize any disruption of this service to the public. The Park Board will continue to maintain these parks and water features for their useful life during the Term, or until such time as the Park Board decommissions, repurposes or replaces either location or equipment. Any Indemnity coverage for activities carried out by the Association at the summer playground and wading pool at Norquay spray park and Bobolink wading pool is provided in accordance with the Indemnity Agreement attached as Appendix E to this Agreement.
- 4. The Fitness Centre is located within the Jointly Operated Facilities and is operated by the Association Any Indemnity coverage for the Fitness Centre is provided in accordance with the Indemnity Agreement attached to this Agreement as Appendix E. All revenue generated by the Fitness Centre is for the benefit of the Association. Revenue generated by the Fitness Centre through the use of the OneCard will be paid to the Association as determined by the reimbursement formula in Appendix B.
- 5. There is a Licensed Childcare space located within the Jointly Operated Facilities, which is operated by the Association. The Association is responsible for this operation Any Indemnity coverage for Licensed Childcare operations is provided in accordance with the Indemnity Agreement attached to this Agreement as Appendix E.

- 6. The Association's affiliated sports groups currently use eleven storage lockers in the Park Board operated ice rink. The Park Board acknowledges and agrees to this use.
- 7. For greater certainty, as of the Effective Date, "Satellite Locations" include the Killarney Tennis Courts, which are owned by the Vancouver School Board. The Park Board books the Killarney Tennis Courts from the Vancouver School Board on behalf of the Association. Any Indemnity coverage for activities carried out at the tennis courts is provided in accordance with the Indemnity Agreement attached to this Agreement as Appendix E.
- 8. Further to Section 7.2 of the Agreement, the Park Board acknowledges and agrees that the Association's current use of Outdoor Areas or other Park Board parks and recreation facilities for Programming purposes and which are bookable through the Park Board internal permitting process or with the prior consent of the Park Board will be preserved. The current permitting process is outlined on the City's website under "Events in Parks". If the current permitting process used by the Park Board and the Association is amended or replaced the Park Board will discuss the possible amendment or replacement with the Association in advance and consider the Association's concerns and suggestions.
- 9. For greater certainty, references in Section 12.3 and 12.5 of the Agreement to the Park Board having "final authority for staffing decisions" and "final authority on staffing decisions" refer only to the Park Board having final authority over staffing decisions with respect to Park Board Personnel. This note is being added for clarification only and does not amend the provisions of the Agreement and should not be construed to modify the interpretation of any other joint operating agreement entered into between the Park Board and a CCA.
- 10. The Park Board recognizes and agrees that the Operation Fee in clause 14.5 is not intended to cause an Association an ongoing deficit in the Association's operating budget. If, at any time during the Term, the Association is in a systemic deficit with its operating budget, the Park Board will work with the Association to consider the issues contributing towards the deficit and the parties will collaboratively develop a sustainable solution to the budgeting issues. The Park Board further acknowledges and agrees that such a solution will not rely on the Association eroding its Retained Earnings.
- 11. The Park Board and Association acknowledge that the Park Board has entered into a Vending Contract with Coca-Cola Bottling Company to allow for the installation of vending machines in certain civic facilities and that, pursuant to that Vending Contract (as it may be amended from time to time), the Park Board is entitled to receive a percentage of revenue generated from such vending machines ("Vending Revenue"). The Park Board acknowledges and agrees that 33% of all Vending Revenue generated by vending machines located in the Jointly Operated Facilities will be provided to the Association within a reasonable period of time after the Park Board receives such Vending Revenue.
- 12. For greater certainty, the Association is an independent society under the Societies Act, and the Association represents that it is not a "member funded society" as defined in the Societies Act nor does it operate a business for the purpose of profit or

gain. The Park Board acknowledges that if the Association has Retained Earnings or a budget surplus that shall not constitute a breach or default under Section 4.2 of the Agreement.

13. The Park Board recognizes that the Association is involved in its community in many ways, and confirms that Section 4.2(c) of the Agreement does not limit the right or ability of the Association to carry out other activities in addition to the provision of Programming and other services within the Community Centre Network.

For greater certainty with regard to Section 4.2(d) of the Agreement, the Park Board acknowledges that the Association is an independent society under the Societies Act and must comply with applicable law and answer to its members. With that recognition of the Association's other legal obligations, the parties wish to clarify that the question of whether the Association has adopted bylaws and/or governance procedures that conflict with the terms of this Agreement or do not permit the Association to carry out its roles and responsibilities under this Agreement, should this question arise, shall be determined on an objectively reasonable standard, is not in the sole discretion of the Park Board, and, if the parties disagree on whether the Association has adopted bylaws and/or governance procedures that conflict with the terms of this Agreement or do not permit the Association to carry out its roles and responsibilities under this Agreement, either party may invoke the dispute resolution mechanism set out in Section 18.1 and proceed to arbitration under Section 18.1A. The parties agree that the interpretation of whether or not a particular bylaw or governance procedure is in conflict with the terms of this Agreement may be determined by arbitration because Section 4.2(d) deals with the Association's internal bylaws and governance procedures. Bylaws or governance procedures adopted by the Association shall in no way exempt the Association from its obligation to comply with Public Policy, as it may be amended, replaced or updated from time to time.

- For greater certainty with regard to Section 6.1 of the Agreement, the Park Board 14. acknowledges that the Association is an independent society under the Societies Act and must comply with applicable law and answer to its members. With that recognition of the Association's other legal obligations, the parties wish to clarify that while the Association is required under this Agreement to adopt and adhere to certain internal policies and standards, as described in the first two paragraphs of Section 6.1 of the Agreement, the question of whether the Association's policies and standards meet the criteria of the first two paragraphs of Section 6.1 of the Agreement, should this question arise, shall be determined on an objectively reasonable standard, and not in the sole discretion of the Park Board, and, if parties disagree on this point, either party may invoke the dispute resolution mechanism set out in Section 18.1 and proceed to arbitration under Section 18.1A. The parties agree that the question of whether the Association's policies and standards meet the criteria of the first two paragraphs of Section 6.1 of the Agreement may be determined by arbitration because the first two paragraphs of Section 6.1 deal specifically with the Association's internal policies and standards.
- 15. The Park Board acknowledges that the Association currently uses Retained Earnings to pay for initiatives which are for the benefit to the community. In addition to the uses specified in Section 14.3 of the Agreement, the Park Board acknowledges and agrees that the Association may make expenditures from Retained Earnings for:

- (a) capital improvements or enhancements to public recreation facilities in the community served by the Association, including parks, playfields or other recreational improvements or facilities;
- (b) bursaries and scholarships for academic achievement, community involvement, community centre involvement, leadership or volunteering;
- (c) events and/or gifts reasonable in the circumstances to recognize and thank individuals who have made a significant volunteer contribution to the operations, programs and services to the community offered by the Association;
- (d) donations to support the provision of public community events or initiatives; and/or
- (e) professional services (including legal services) relating to: (i) the implementation and administration of the Agreement; (ii) Programming, services and community building initiatives offered by the Society at or from the Jointly Operated Facilities; (iii) planning for physical improvements to or capital projects within the Community Centre Network; or (iv) capacity building initiatives of the Association for the purposes of enhancing the provision of services to the community.

The Park Board acknowledges that these initiatives are of benefit to the community and agrees that they are an appropriate use for Retained Earnings. Furthermore, for greater clarity with regard to Section 14.3(e) of the Agreement and Retained Earnings, the parties understand and agree that the Association must incur certain expenses in order to provide the Programming and other services within the Community Centre Network and, as set out in Section 14.3(e), the Association may spend or use its Retained Earnings as an operating contingency for such purposes.

16. The Park Board and the Association agree that, during the last year of the Term, the System-wide Planning Sessions will be used by the parties to discuss the ongoing relationship of the parties after the Term and what changes may be required to this Agreement prior to the parties entering into a new joint operating agreement for a new term, should both parties wish to continue joint operation of the Jointly Operated Facilities after such discussions. If the parties consider it prudent, they may schedule additional sessions in addition to the System-wide Planning Sessions to ensure fulsome good faith discussions about the continuing relationship of the parties after the Term.

For the purposes of this section, the parties agree that "fulsome" will mean "abundant and encompassing of all aspects".

17. For the purposes of Section 20.1 of the Agreement, the parties agree that if the Park Board determines that the Association is or has been in sustained breach of the Agreement or if the Park Board holds reasonable concerns regarding the Association's ability to operate the new space in a safe, professional and compliant manner such that the new space is not automatically included in the scope of the Jointly Operated Facilities, the decision about whether or not to include the new space within the

scope of the Jointly Operated Facilities shall be determined on an objectively reasonable standard and on an objectively reasonable basis. If the Association disagrees with whether the Association is or has been in sustained breach of this Agreement or whether the decision to include or exclude the new space from the scope of the Jointly Operated Facilities has been determined on an objectively reasonable standard and on an objectively reasonable basis, then the Association may invoke the dispute resolution mechanism set out in Section 18.1 of the Agreement and, if the matter is not resolved after proceeding through the dispute resolution process, the Association may elect to proceed to arbitration under Section 18.1A.

For greater clarity and for the purposes of Section 20.1 of the Agreement, a "sustained" breach shall mean a breach not remedied (cured) in accordance with the timelines and processes set out in Section 21.1 of the Agreement (which timelines and processes, for the purpose of Section 20.1 of the Agreement, shall not require a notice of termination to initiate the timelines and processes, and do not result in the termination of the Agreement) and may include a breach that is not "material" in nature. For any breach that occurs after the Effective Date, the Park Board will notify the Association in writing about the breach of the Agreement within a reasonable period of time after discovery, and provide a reasonably detailed explanation of the breach.

For greater clarity and for the purposes of Section 20.1 of the Agreement, if, at any time after the Effective Date, the Association acts in such a manner as to cause a reasonable concern for the Park Board about the ability of the Association to operate recreational space in a safe, professional and compliant manner, the Park Board will notify the Association in writing about the action and the resultant concern within a reasonable period of time after discovery, and provide a reasonably detailed explanation of the concern so that the Association has the opportunity to remedy the concern within a reasonable period of time.

For the purposes of Section 20.3 of the Agreement, the parties agree that if the Park 18. Board determines that the Association is or has been in sustained breach of the Agreement or if the Park Board holds reasonable concerns regarding the Association's ability to operate all or any component of the replacement facility in a safe, professional and compliant manner such that this Agreement is not extended to all or any component of the replacement facility, the decision about whether or not to extend this Agreement to all or any component of the replacement facility shall be determined on an objectively reasonable standard and on an objectively reasonable basis. If the Association disagrees with whether the Association is or has been in sustained breach of this Agreement, or whether the decision to extend this Agreement to all or any component of the replacement facility has been determined on an objectively reasonable standard and on an objectively reasonable basis, then the Association may invoke the dispute resolution mechanism set out in Section 18.1 of the Agreement and, if the matter is not resolved after proceeding through the dispute resolution process, the Association may elect to proceed to arbitration under Section 18.1A.

For greater clarity and for the purposes of Section 20.3 of the Agreement, a "sustained" breach shall mean a breach not remedied (cured) in accordance with the timelines and processes set out in Section 21.1 of the Agreement (which timelines and

processes, for the purpose of Section 20.3 of the Agreement, shall not require a notice of termination to initiate the timelines and processes, and do not result in the termination of the Agreement) and may include a breach that is not "material" in nature. For any breach that occurs after the Effective Date, the Park Board will notify the Association in writing about the breach of the Agreement within a reasonable period of time after discovery, and provide a reasonably detailed explanation of the breach.

For greater clarity and for the purposes of Section 20.3 of the Agreement, if, at any time after the Effective Date, the Association acts in such a manner as to cause a reasonable concern for the Park Board about the ability of the Association to operate recreational space in a safe, professional and compliant manner, the Park Board will notify the Association in writing about the action and the resultant concern within a reasonable period of time after discovery, and provide a reasonably detailed explanation of the concern so that the Association has the opportunity to remedy the concern within a reasonable period of time.

The Park Board and the Association acknowledge that as a condition of entering into the Agreement, the Association withdrew its part of the lawsuit against the Park Board (Hastings Community Association et al. v. the Park Board, Vancouver Registry No. 136262) and the Association and the Park Board entered into a mutual release in respect of that litigation (the "Mutual Release"). The Park Board and the Association agree that the Mutual Release contains certain terms which modify sections 20.1 and 20.3 of the Agreement and sections 17 and 18 of this Appendix C with respect to known breaches which predate this Agreement. The Park Board and the Association agree that those sections of the Mutual Release form part of this Agreement, notwithstanding the language of section 24.9 of the Agreement.

The Park Board confirms that the termination provisions in Section 21.1 apply only in 19. the case of a sustained, material, un-remedied breach of the Agreement. If the Association disagrees whether or not a breach meets these criteria, they may invoke the dispute resolution mechanism set out in Section 18.1 of the Agreement. The Park Board further confirms that the Association has a cure period of 60 days (as more specifically set out in Section 21.1) to remedy such breach (except in the case of emergency) and that if more than 60 days is required to remedy the breach (using reasonable resources and acting diligently), then the Association will not be considered to be in breach after 60 days and for such longer period as is required to remedy the breach, provided the Association is making reasonable and diligent efforts to remedy the breach in a timely manner (all as more specifically set out in Section 21.1). For additional clarity, the Park Board confirms that "reasonable and diligent efforts" must be objectively reasonable and it is not in the sole discretion of the Park Board whether efforts to remedy a breach are "reasonable and diligent". If the parties disagree with whether efforts have been reasonable and diligent, either party may invoke the dispute resolution mechanism set out in Section 18.1. With regard to Section 21.1(e) of the Agreement, the parties wish to confirm that "continuously and diligently" must be objectively reasonable and it is not in the sole discretion of the Park Board whether efforts to remedy a breach are being made "continuously and diligently". If the parties disagree with whether efforts have been continuous and diligent, either party may invoke the dispute resolution mechanism set out in Section 18.1. Further with regard to Section 21.1(e), the parties wish to clarify that the Park

- Board's right to terminate the Agreement with no further notice required occurs only after the initial notice periods contemplated by Section 21 have passed.
- 20. In the interest of clarity the parties wish to confirm that, at all times during the Term including any Renewal Term, the Agreement shall remain in full force and effect during any dispute resolution mechanisms as set out in the Agreement.
- 21. The parties agree that the negotiations to enter into the Agreement have been carried out between the CCAs and Park Board but that the final approval of the body of the Agreement was given by the Park Board Commissioners on behalf of the Park Board. With that in mind, and for greater certainty with regard to the administration of the Agreement, the Park Board acknowledges that, while authority to administer the Agreements rests with the General Manager of the Park Board, the General Manager of the Park Board will not issue a notice of termination pursuant to the Agreement unless it is authorized by the Park Board Commissioners.
- 22. The Park Board agrees that if the Park Board agrees to include wording in the appendices of another CCA that would have the effect of modifying or amending a section of the body of the Agreement, the Park Board agrees that it will provide such wording to the Association and such wording may be added to this Appendix C at the election of the Association. The foregoing will not apply in the case of modifications to the appendices of a CCA which are intended to address current practice or unique operations of a particular CCA.

This Appendix C may be adjusted by the parties at any time, upon mutual agreement, as practices or operations of the parties may change over time.

APPENDIX D

Operational Details for ActiveNet

The parties agree the ActiveNet will be used at the Jointly Operated Facilities in accordance with the following practices:

- (a) ActiveNet will be used at the Jointly Operated Facilities for administrative functions, including to process, record, and store registration information for Programming, to process and record payments of Centrally Processed Revenue, customer account inquiries, and front desk and customer service administration. The functions and modules currently processed through ActiveNet (and previously processed through the Safari system) will continue to be processed through ActiveNet, consistent with the existing practices of the parties;
- (b) the City/Park Board will collect Centrally Processed Revenue on behalf of the Association through ActiveNet;
- (c) payment in all forms (cheques, cash, credit, debit) for Programming, services, rentals, childcare, and any other activity provided by the Associations and administered with the support of Park Board or City staff will continue to be processed through ActiveNet, as is currently the case;
- (d) the Park Board/City will remit the Centrally Processed Revenue to the Association, subject only to Authorized Deductions, on the current bi-weekly schedule of payments unless a different schedule of payments is agreed to by the parties;
- (e) except for the Authorized Deductions, the Park Board/City will not withhold payment of any Centrally Processed Revenue to the Association through ActiveNet for any reason without the Association's written consent;
- (f) within the limits of ActiveNet functionality, the City/Park Board will provide to or make available to the Association timely ActiveNet reports on Centrally Processed Revenue, payments, deductions, refunds, and taxes by providing ActiveNet logins and training to Association and/or by scheduling reports to be automatically sent to Association;
- (g) the Association will meet with the Park Board and/or City to resolve non-technical issues within 15 business days of a request by the Park Board;
- (h) the City/Park Board meet with the Association within 15 business days of a request by the CCAs to resolve non-technical issues and make changes to resolve the issues raised by the CCAs or any of them, unless the Park Board and/or City provides written reasons as to why the requested change is not practically feasible;
- (i) the Park Board/City will continue to provide technical support to the Association for ActiveNet and work with ActiveNet to ensure technical support is available to resolve any issues;

- (j) the parties will meet and provide regular and ongoing feedback to each other on how ActiveNet is working, reporting needs, financial and bookkeeping needs, and other topics;
- (k) either party will meet and review the payment schedule for Centrally Processed Revenue within a reasonable period of time, if requested by the other party;
- (l) all receipts generated through ActiveNet will bear the name of the Association and the Park Board in equal prominence, except for debit and credit card slips, which bear the name of the account and the name of the Jointly Operated Facilities;
- individuals registering for or purchasing Programming through ActiveNet in person or online will be able to, in person and online, become a member of the Association offering the Programming;
- (n) The insufficient fund ("NSF") fees, if any, currently charged by Association will be maintained. The City/Park Board will collect and retain those fees, except in instances where a particular NSF fee has been waived by an Association. Each Association may waive an NSF fee on a case by case basis according to the financial needs of the individual incurring the NSF fee;
- (o) The Park Board and the City agree that any limitations of ActiveNet shall not constitute grounds for the Park Board or the City to interfere with the current Association practices for determining and implementing fees to be charged to patrons for Programming, services, rentals, childcare, and any other activity provided by the Association; and
- (p) The City and Park Board agree that, if the Association wishes to explore the option of no longer using ActiveNet for childcare administration, the City and/or Park Board will meet with the Association to discuss the implications of the Association no longer administering childcare payments and childcare registration in ActiveNet.

APPENDIX E

Indemnity Agreement



INDEMNITY AGREEMENT

THIS AGREEMENT made as of 2000 F

BETWEEN: KARDARNEN COMMINIEN CHATTER SOCIETY

on behalf of itself and affiliated organizations as hereinafter defined

(hereinafter called (he "Society")

OF THE FIRST PART

AND:

BOARD OF PARKS AND RECREATION having offices at 2099 Beach Avanue, Vancouver, Brilish Columbia, V6G 1Z4

OF THE SECOND

PART AND:

CITY OF VANCOUVER having offices at 453 West 12th Avenue Vancouver, British Columbia, V5Y 1V4

(hereinafter called the "City")

OF THE THIRD PART

For the purposes of this Agreement "affiliated organization" means an organization or individual who develops and operates a recreational program at the request of and under the direction and supervision of the Society, and when so acting, shall be deemed to be an agent of the Society. An affiliated organization must have its programs approved for indemnification purposes using the process outlined in this agreement and indemnification is subject to the conditions outlined in this agreement conditions outlined in this agreement.

This Agreement replaces indemnity agreements previously entered into between the Parties.

WHEREAS the Board has, with the authorization of Council of the WITERCAS the poeto has, with the authorization of Council of the City, decided to provide recreational programs for the recreation, comfort and enjoyment of the public;

AND WHEREAS the Society has offered and agreed to develop and

en (a) lead a ward





provide such recreational programs, either directly or by an affiliated organization;

AND WHEREAS, as a condition of developing and providing such recreational programs, the Society has requested that the Board and the City indemnify its officers, employees, directors, and volunteers of the affiliated organizations, against any claims, demands and actions erising out of the provision recreational programs developed and provided at the request and approval of the Board.

THE BOARD AND THE CITY REREBY AGREETHAT the City will indemnify, to a maximum amount of five million dollars per occurrence, the Society, or affiliated organization and their current or former officers, directors, employees and volunteers, against any claim, demand or action brought against it or its current or former officers, employees, directors or volunteers in connection with the development or provision of a recreational program PROVIDED THAT such recreational program has been approved in writing by the General Manager of the Board and the General Manager of the Board has agreed in writing to provide indemnity in connection with such recreational program. Any approval or agreement to indemnify shall be subject to the following conditions:

- (a) The City Manager of the City or the General Manager of the Board may at any time instruct the Society to cease operation of the recreational program.
- (b) Indemnity will not be provided where the claim, demand or action arises out of a malicious act or amission.
- (c) Indemnity will be provided only where the program is being developed and provided in accordance with the approval given and any guidelines or instructions subsequently set by the Board through its General Manager and not contrary to any direction given by Council of the City. By submitting a recreational program for approval the Society and affiliated organization shall be taken to have agreed to comply with such guidelines or instructions.
- (d) The Society or affiliated organization will provide to the Board and the City, upon request, any information with respect to a program for which indemnity has been requested or been provided.
- (e) Either the Board or the City may, on 30 days william notice to the Society or an affiliated organization, cancel and revoke their agreement to Indemnify contained herein, and where such notice is given, then on the expiration of 30 days, neither the Board or the City has any obligation to indamnify the Society or affiliated organization or any of its officers, employees or volunteers in respect of any act or omission occurring after cancellation.

The notice referred to herein may be given by mailing or delivering it to the address of the Society which has been given to the Board for such purpose or by posting it in the building or facility in or from which the program is operated. Where notice is given by posting it, the period of Notice shall commence 7 days after it has been posted.

- (f) The Society or affiliated organization and their officers, employees, directors and volunteers will do nothing to prejudice the defence of any claim, demand or action and full co-operate with the City and the Board in the defence of any claim, demand or action.
- (g) The City shall have the conduct of, and bear the costs associated with, the defence to any claim, demand or action in respect of which indemnity is

phidiadewood

2.

provided (and therefore no financial consequence exists for the indemnified individual or the Society) and shall have the unfettered right to make all decisions in respect of indemnified claims or actions, including the right to settle or compromise the claim, demand or action. If legal counsel for the City determines that there is a divergence of interest between the City and any party covered by the indemnity, then independent legal counsel would be obtained and the costs of such would be paid by the City as part of the indemnity.

(h) Where an indemnity is provided pursuant to this Agreement, the City and the Board agree that the affiliated organization developing or providing the approved recreational program shall be deemed to be a party to this Agreement.

IN WITNESS W HEREOF the parties have hereto set their hands and seals or caused their corporate seals to be affixed under the hands of their proper officers duly authorized in that behalf, as the case may be, as of the day and year first above written.

The Common Seal of	
was hereunto affixed) in the presence of:	
let. K.c.c.s.	
Authorized Signatury	c/s
Authorized Signatory Targemen.	•
The Board of Parks and Recreation by:	C/S
j)
Chairman	
Drunner.)
General Manager	
The Common Seal of the CITY OF VANCOUVER was hereunto affixed	C/S
in the presence of:)
Authorized Signaldry	

philink.ard

:

MUTUAL RELEASE

AMONG:

KILLARNEY COMMUNITY CENTRE SOCIETY ("CCA")

AND:

THE VANCOUVER BOARD OF PARKS AND RECREATION of the CITY OF VANCOUVER (the "Park Board") and the CITY OF VANCOUVER (the "City")

WHEREAS:

- 1. The CCA is a society governed by the Societies Act, S.B.C. 2015, c. 18;
- 2. The Park Board is a statutory elected body under the *Vancouver Charter*, S.B.C. 1953, c. 55.
- 3. The CCA and the Park Board are parties to a Joint Operating Agreement (the "Current JOA") dated June 1, 1979;
- 4. The CCA commenced an action against the Park Board in the Supreme Court of British Columbia, Action No. S-136262, Vancouver Registry on August 20, 2013 (the "Action");
- 5. The CCA and the Park Board wish to resolve the Action and any disputes under the Current JOA and enter into a new Joint Operating Agreement dated _______, 2018 (the "New JOA").

THE PARTIES AGREE AS FOLLOWS:

FOR AND IN CONSIDERATION of each party waiving their legal costs incurred in the above-captioned matter; the Park Board entering into the New JOA with the CCA in the form previously approved by the Park Board; and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged,

(a) THE CCA DOES HEREBY, WITHOUT QUALIFICATION OR LIMITATION, REMISE, RELEASE AND FOREVER DISCHARGE THE PARK BOARD AND THE CITY, including their respective heirs, executors, administrators, servants, agents, employees, elected officials, solicitors, directors, officers, officials, successors and assigns, subsidiaries, parents and affiliated companies, reinsurers, or insurers, as the case may be, from any and all causes of action, claims, suits, demands, damages, costs and expenses of any nature or kind whatsoever, whether at law or in equity, known or unknown, which each of the parties to this Mutual Release, or their respective heirs, executors,

administrators, servants, employees, agents, solicitors, directors, officers, officials, successors and assigns, subsidiaries, parents and affiliated companies, reinsurers, or insurers, as the case may be, now have or at any time hereafter can, shall or may have in any way resulting or arising from or in connection with:

- (i) the matters at issue in the Action;
- (ii) the Current JOA;
- (iii) any past conduct, act or omission of the Park Board or the City that occurred prior to the date of signing this Mutual Release; and
- (b) THE PARK BOARD AND THE CITY DO HEREBY, WITHOUT QUALIFICATION OR LIMITATION, REMISE, RELEASE AND FOREVER DISCHARGE THE CCA, including their respective heirs, executors, administrators, servants, agents, employees, solicitors, directors, officers, officials, successors and assigns, subsidiaries, parents and affiliated companies, reinsurers, or insurers, as the case may be, from any and all causes of action, claims, suits, demands, damages, costs and expenses of any nature or kind whatsoever, whether at law or in equity, known or unknown, which each of the parties to this Mutual Release, or their respective heirs, executors, administrators, servants, employees, agents, solicitors, directors, officers, officials, successors and assigns, subsidiaries, parents and affiliated companies, reinsurers, or insurers, as the case may be, now have or at any time hereafter can, shall or may have in any way resulting or arising from or in connection with:
 - (i) the matters at issue in the Action;
 - (ii) the Current JOA;
 - (iii) any past conduct, act or omission of the CCA that occurred prior to the date of signing this Mutual Release.

IT IS EXPRESSLY UNDERSTOOD AND AGREED that for the purposes of sections 20.1 and 20.3 of the New JOA and sections 17 and 18 of Appendix "C" to the New JOA, the Park Board, in determining whether the CCA will operate additional space in the Entire Facility (as that term is defined in the New JOA) or operate replacement facilities to the Jointly Operated Facilities (as that term is defined in the New JOA) under those aforementioned sections of the New JOA, hereby waives the right to consider any past breaches by the CCA of the terms of the Current JOA that the Park Board had actual knowledge of prior to entering into the New JOA.

IT IS FURTHER EXPRESSLY UNDERSTOOD AND AGREED that the settlement is a compromised settlement of disputed claims and the consideration referred to herein shall not be construed as an admission on the part of any of the parties to this Mutual Release, by whom liability is expressly denied.

IT IS FURTHER EXPRESSLY UNDERSTOOD AND AGREED that none of the parties to this Mutual Release will make any further claims or take any further proceedings in respect of the matters which are the subject of this Mutual Release against any other party, person, company, corporation, or other legal entity who might claim contribution or indemnity from any of the

parties to this Mutual Release, or their respective heirs, executors, administrators, officers, directors, employees, servants, agents, predecessors, successors, assigns, subsidiaries, parents and affiliated companies, reinsurers or insurers, as the case may be.

IT IS FURTHER EXPRESSLY UNDERSTOOD AND AGREED and all parties to this Mutual Release ACKNOWLEDGE that the facts in respect of which this Mutual Release are made may prove to be other than or different from the facts in that connection now known by any of the parties or any one or more of them or believed by any of them to be true. All parties to this Mutual Release EXPRESSLY ACCEPT AND ASSUME the risk of the facts being different and agree that all of the terms of this Mutual Release shall be in all respects effective and not subject to termination or rescission by discovery of any difference in the facts.

THE PARK BOARD AND THE CCA do hereby authorize and instruct each of their respective solicitors to consent to the dismissal or discontinuance of the Action, at the Park Board's election, without costs to any party and each of them also consent to the release of the injunction Order pronounced by Mr. Justice Bowden on January 17, 2014 (the "Injunction Order"), as it pertains to the Park Board and the CCA. The effect will be that the CCA shall no longer be a party to the Action and cannot enforce the Injunction Order. For greater certainty, the terms of this Mutual Release shall have no effect on the Park Board and the remaining parties to the Action, other than those parties that have entered into a mutual release with the Park Board and the City, from continuing to pursue individual claims or defences in the Action or from enforcing the Injunction Order against the remaining parties to the Action.

BY SIGNING THIS AGREEMENT, THE PARTIES acknowledge that they expressly understand and agree with the terms herein and both represent and declare that they have read this Mutual Release, have been advised by their solicitors before entering into the agreement herein contained and confirm that this Mutual Release contains the entire agreement between them and that the terms of this Mutual Release are contractual and not merely a recital.

THIS DOCUMENT MAY BE EXECUTED in counterparts and when all of the counterparts have
been executed, the document shall constitute a full and binding Mutual Release.

Dated at <u>Vancouver</u> ,	, the	<u>15</u> d	lay of _	May	, 2018
-----------------------------	-------	-------------	----------	-----	--------

SIGNED, SEALED & DELIVERED in the presence of	WTupung (mod
Signature of Witness	1 Vo Injung / presi
Name: AINSUE KWAN	KILLARNÉY COMMÚNITY CENTRE SOCIETY
Address: 5880 MC Kinnon St.	
VANCOUVER BC USR4E3	WTupling Gnest
Occupation: OFFICE ADMIN ASST.	Print Name of Authorized Signatory
SIGNED, SEALED & DELIVERED in the presence of: Signature of Witness	<u> Mahabah</u>
Name: KARIN GRUBB	VANCOUVER BØARD OF PARKS AND RECREATION
Address: 453 WEST 12TH AVENUE) Address: VANCOUVER, B.C. VSY 1V4) LAWYER)	Malcolm Bromley Print Name of Authorized Signatory
Occupation:	, -

SIGNED, SEALED & DELIVERED in the presence of:))
W	
Signature of Witness	rances . Come
Name: KARIN GRUBB	CITY OF VANCOUVER CONNELL
Address: 453 WEST 12 TH AVENUE VANCOUVER, B.C. V5Y 1V4 LAWYER) 453 WEST 12th AVENUE) VANCOUVER BC V5Y 1V4) BARRISTER & SOLICITOR
E WITH STATE	Print Name of Authorized Signatory
Occupation:	<i>)</i>)